On the family origins of human capital formation: Evidence from donor children.

Petter Lundborg, Erik Plug, and Astrid Würtz Rasmussen*

Abstract

We introduce a novel strategy to study the intergenerational transmission of human capital, net of genetic skill transfers. For this purpose, we use unique data on children conceived through sperm and egg donation in IVF treatments in Denmark. Because the assignment of donors is not selective, the intergenerational human capital estimates allow for a causal nurture interpretation. Once we take account of genes, we find that only the education of mothers matters: the association between mother's education and child test scores is significant and large, whereas the association between father's education and child test scores is insignificant and practically zero.

^{*}Plug: Amsterdam School of Economics, University of Amsterdam, Roetersstraat 11, 1018 WB Amsterdam, The Netherlands (email: e.j.s.plug@uva.nl). Lundborg: Department of Economics, Lund University, Tycho Brahes Väg 1, 223 63 Lund, Sweden (email: petter.lundborg@nek.lu.se). Rasmussen: Department of Economics and Business Economics, Aarhus University, Fuglesangs Allé 4, 8210 Aarhus, Denmark (email: awr@econ.au.dk). Data in this article are, in part, taken from IVF registers, held by the Danish Health Data Authority (DHDA). DHDA is not responsible for the analysis or interpretation of the data presented. We thank seminar and conference participants in Aarhus, Amsterdam, Antwerp, Barcelona, Berlin, Bonn, Copenhagen, Gothenburg, Göttingen, Groningen, Helsinki, Konstanz, London, Lund, Luxembourg, Lyon, Madrid, Marseille, Milan, Munich, Odense, Reus, Rotterdam, Stavanger, Tilburg, Trondheim, Uppsala, Venice, Vienna, and Zurich for their comments and suggestions. Financial support from the Institute for Evaluation of Labour Market and Education Policy (IFAU), Swedish Research Council for Health, Working Life and Welfare (FORTE), and the Dutch Research Council (NWO) is gratefully acknowledged. The authors declare that they have no relevant or material financial interest that relate to the research described in this paper.

1 Introduction

Why do better educated parents have better educated children? This question has attracted continuous attention from social scientists for well over a century. Their polar explanations are one of nurture, and one of nature. It is nurture if better educated parents provide a more advantageous environment for their children to do well in school. It is nature if better educated parents have certain genetic advantages that they pass on to their children. Of course, any intermediate explanation is possible too, and arguably more likely.

Previous adoption and twin studies seem to settle on nurture being (somewhat) less important than nature for the formation and intergenerational transmission of human capital (Taubman 1976, Plug and Vijverberg 2003, Plug 2004, Björklund, Lindahl, and Plug 2006, Björklund, Jännti, and Solon 2005, Sacerdote 2002 2007, Cesarini 2010, Cesarini and Visser 2017). There is, however, much uncertainty about the accuracy of these nurture and nature estimates; as noted in recent literature surveys, the adoption and twin strategies used to isolate nurture from nature influences often suffer from identification problems that bias results against the nurture explanation (Björklund and Salvanes 2011, Black and Devereux 2011, Holmlund, Lindahl, and Plug 2011, and Sacerdote 2011).

In this article, we introduce a novel identification strategy to more credibly identify the nurture effect in the intergenerational transmission of education. In particular, we exploit that some children are genetically unrelated to one of their rearing parents because they are conceived through sperm or egg donors in in vitro fertilization (IVF) treatments. Sperm donation refers to fertilization of the mother's egg with the sperm of an anonymous donor man. Resulting children are genetically related to the mother but not the father. Egg donation is like sperm donation in that the children are genetically related to the father but not the mother. In this IVF context, we identify the nurture effect by estimating how the educational outcomes of donor children relate to the educational outcomes of their genetically unrelated parents. Because the assignment of donors is not selective, we can give the corresponding intergenerational mobility estimates a causal nurture effect interpretation, one that captures the nurturing effect of both prenatal and postnatal environments.¹

¹In case nurture and nature interact, the nurture effects we estimate represent the average of different nurture effects, which then vary with the innate skills of donor children. When we explore nature-nurture interactions, however, we find little evidence that these interactions matter much for child test score outcomes.

We use several administrative registers to compile our primary sample of IVF children born in Denmark (between 1994-2007) with information on their donor status (conceived through sperm or egg donation), on various test score outcomes from nationwide standardized tests taken throughout their primary and lower secondary school years, and on the educational and labor market characteristics of their parents. This sample allows us to estimate intergenerational associations for donor and nondonor children. In addition, we use the same registers to compile two validity samples of children: one of adopted children, and one of all other children. These samples together allow us to compare intergenerational associations between donor and adopted children (to better assess the role of prenatal and postnatal environments) and between nondonor and all other children born in Denmark in this period (to better assess the generalizability of our findings).

To preview our main results, we find no evidence that the education of fathers matters for their children's test scores, once we take account of their genes. In donor families where children are genetically unrelated to the father, the association between paternal education and child test scores is statistically insignificant and close to zero. By contrast, we find strong evidence that the education of mothers matters, also net of their genes. In donor families where children are genetically unrelated to the mother, we find that children with better educated mothers perform much better on standardized achievement tests. The association between maternal education and child test scores is significant and large, and almost as large as the association we find for mothers and genetically related children.

It is important to know whether these nurture effects, or absence thereof, gathered in donor families generalize to more representative families. We identify two external validity risks: family heterogeneity in intergenerational skill effects and differences in parenting styles. Exploring the empirical relevance of these risks in a setting where children in donor families are genetically related to only one of their parents, we find no evidence that intergenerational skill transmission between parents and genetically related children differs in donor and representative families, nor that parents spend different amounts of time with their genetically unrelated children (in infancy and early childhood). We therefore believe that the nurture effects taken from donor families carry over to more representative families.

The paper unfolds as follows. Section 2 provides the literature background, and lists the main contributions of our study. Section 3 presents the institutional

context of IVF treatments in Denmark and the administrative data. Section 4 describes how donors are assigned to IVF treated families. Section 5 introduces the novel strategy to identify the nurture influences of parental education in intergenerational mobility models using donor and nondonor IVF children. Section 6 presents our main set of results. Section 7 compares our results to those obtained using representative and adoption samples. Section 8 concludes.

2 Previous Literature

There is an active literature concerned with estimating the nurturing effect of family background on the skills and educational outcomes of children. See Björklund and Salvanes (2011), Black and Devereux (2011), Holmlund, Lindahl, and Plug (2011), and Sacerdote (2011) for recent literature reviews on the topic. In isolating nurture from nature influences, most studies estimate education associations of either sibling pairs or parent-child pairs using genetically informative samples of twins and adoptees. In this section, we will first summarize the nurture findings from previous sibling and intergenerational studies, before we turn to the main contributions of our study.

Sibling Studies

One line of studies relies on sibling associations. With different sibling pairs sharing different combinations of common genes and environment, researchers have used behavioral genetic models to decompose the overall educational outcome variation into nature and nurture components. The nurture component represents the impact of some latent family background component that captures all the environmental factors shared by siblings.

Studies that use twins identify the nature component from sibling association differences between identical and fraternal twins, which is then used to recover the nurture component. Studies that use adoptees identify the nurture component from sibling associations of either adopted siblings or adopted and non-adopted siblings. These studies generally find that nurture matters, explaining about 10 to 45 percent of the overall variation in the educational attainment.² These nurture estimates, however, rely on rather controversial

 $^{^{2}}$ We have taken sibling associations in education reported elsewhere and constructed a comprehensive set of nurture estimates. With twins, the nurture estimates range from 10 percent (Miller, Mulvey, and Martin 1995), 25-35 percent (Taubman 1976, Jencks and Brown 1977, Lykken et al. 1990, Isacsson 1999, Cesarini 2010), to 45 percent (Ashenfelter and

model assumptions regarding the representativeness of twins and adoptees, siblings (not) affecting one another, gene-environment independency, similarity in (parental) treatments, random partner choice (in case of twins), and random assignment to families (in case of adoptees). Any conclusion based on such nurture estimates should therefore be treated cautiously.

Intergenerational Studies

Another line of studies relies on intergenerational associations. With different parents providing different combinations of genes and environment to their children, researchers have used regression models to link the educational characteristics of parents and children, after taking account of genetic skill transfers. The corresponding estimate expresses the nurturing impact of parental education, which in our case is the causal effect of parental education and any other environmental factor that is correlated with it.

Studies that use identical twin parents identify the nurturing effect from within-twin regressions linking the educational differences of twin parents to the educational differences of their children.³ Studies that use adoptees identify the nurturing effect from simple cross-sectional regressions linking the educational outcomes of adoptive parents to the educational outcomes of their adopted children. Like the sibling association studies, these studies also find that nurture matters, explaining about 30 percent of the overall intergenerational association in education for mothers, and about 60 percent for fathers.⁴

Krueger 1994). With adoptees, the nurture estimates equally vary and range from 10 percent (Scarr and Weinberg 1994), 20 percent (Lichtenstein, Pedersen, and McClearn 1992, Cesarini 2010), to 45 percent (Teasdale and Owen 1984). Recent adoption studies that account for random assignment of adoptees to families find that nurture explains about 15 percent of the overall education variability (Sacerdote 2007, Fagereng, Mogstad, and Rønning 2018).

³This nurturing interpretation of within-twin estimates relies on an older argument that within-twin strategies used to identify the causal impact of parental education fail to account for all the relevant non-heritable ability differences between twins. These strategies therefore produce within-twin estimates that capture the impact of parental education and all the unshared non-heritable abilities that are correlated with it (Griliches 1979).

⁴Analogous to the sibling associations we summarized earlier, we have collected intergenerational associations in education from twin and adoption studies and expressed the corresponding nurture effect estimates as fractions of the overall (cross-sectional) intergenerational associations in education. With twin parents, we find that most fraction estimates range from 25 to 40 percent for mothers, and from 30 to 75 percent for fathers (Behrman and Rosenzweig 2002, Bingley, Christensen, and Jensen 2009, Holmlund, Lindahl, and Plug 2011, Pronzato 2012, Amin, Lundborg, and Rooth 2015). With adoptees, these fraction estimates are somewhat higher, ranging from 50 to 65 percent for mothers, and from 65 to 85 percent for fathers (Dearden, Machin, and Reed 1997, Sacerdote 2000, Plug 2004, Holmlund, Lindahl, and Plug 2011). Recent adoption studies that more carefully account for the random assignment of adoptees to families report smaller fraction estimates of about 30 percent for mothers, and

While these regression models provide nurture estimates that are easier to interpret than those provided by behavioral genetic decomposition models, there are limitations that may bias these nurture effect estimates downwards. The main limitation in twin studies relates to the non-heritable traits that twins share. If the differencing of twins not only differences out all the heritable traits but also some of the non-heritable traits that twins share, as Griliches (1979) and others have intermittently argued, the within-twin nurture effect estimates likely understate the nurturing influence of parental education. And similarly, the main limitation in adoption studies relates to the early childhood environment that adoptees may miss. If there is an important role for prenatal and early childhood conditions in explaining child outcomes, as Almond, Currie, and Duque (2018) and others have repeatedly found, the nurture effect estimates taken from adoptees will also understate the true impact of nurture on those outcomes.⁵

Our Contributions

Our study on children conceived through sperm and egg donations, when viewed as embryo adoptions, most closely relates to the recent adoption studies, which account for non-random assignment and investigate how educational and wealth outcomes of adopted children relate to the educational outcomes of their rearing parents (Björklund, Lindahl, and Plug 2006, Sacerdote 2007, Hægeland et al. 2010, Holmlund, Lindahl and Plug 2011, Fagereng, Mogstad, and Rønning 2018, Black et al. 2019). Our study is complementary to these adoption studies in two important ways. First, these adoption studies estimate nurture effects on samples of children who are adopted during early childhood (up to six to eighteen months). Our study estimates nurture effects on samples of donor children who are transferred into the womb 3 to 5 days after a successful fertilization. This means that, unlike the adoption studies, our nurture effect estimates capture prenatal and very early childhood influences. Second, the adoption studies rely on parents and children that bear little resemblance to any other sample of (representative) parents and children.⁶ Our study uses samples of donor chil-

about 45 percent for fathers (Björklund, Lindahl, and Plug 2006, Sacerdote 2007, Black et al. 2019).

 $^{^{5}}$ Other limitations relate to the larger impact of measurement error on within-twin estimates, the non-random assignment of adopted children into adoptive families, and the lack of representativeness of samples of twins and adoptees.

 $^{^{6}}$ It is difficult to compare adoption and non-adoption families for a number of reasons. Adoptees are less comparable in that they are separated from their birth parents, possibly

dren born and raised by IVF parents treated with either eggs from other IVF treated mothers or sperm from Danish sperm donors (with arguably comparable traits).⁷ This means that our study has a greater representation than the adoption studies.

Compared to these adoption studies, however, our study also comes with two limitations. First, IVF treatments based on donors are quite rare which means that we work with relatively small samples.⁸. Second, IVF treatments based on donors are quite recent interventions which means that most donor children are too young to measure their performance in terms of realized educational attainment and labor market outcomes. Instead, we work with intermediate school outcomes and measure the children's performance in terms of test score outcomes from national tests taken in primary and lower secondary education. Comparable test scores taken in primary education are often found to be strong predictors for outcomes that are realized later in life such as final exam scores, educational attainment and labor market earnings (Beuchert and Nandrup 2018, Woessmann 2018).⁹

with traumatizing effects (Brodzinski 1987). Adoptees are also less comparable in that they are assigned to rearing parents that are very different from themselves. International adoptees (including Korean-born adoptees) look distinctively different from their rearing parents. National adoptees may look more similar to their parents, but often have distinctively different backgrounds. Björklund, Lindahl, and Plug (2006) document that Swedish-born adoptees are mostly born in less-advantaged families but placed in more-advantaged families.

⁷Nonexperimental evidence suggests that donor-treated parents under less restrictive donor assignment rules tend to choose donors who resemble the (infertile) partner (Nielsen, Pedersen, and Lauritsen 1995).

⁸This small sample limitation is shared with many twin and adoption studies on the topic. Examples of the earlier intergenerational twin and adoption studies are Behrman and Rosenzweig (2002), Sacerdote (2002), and Plug (2004). Behrman and Rosenzweig (2002) identify the intergenerational transmission of human capital in the Minnesota Twin Registry (MTR) by taking differences between identical twin parents with different levels of education; this is the case for 66 pairs of twin fathers and 87 pairs of twin mothers. Sacerdote (2002) identifies the nurturing impact of father education on child test scores and college attendance using 81 adoptees taken from the British National Child Development Survey (NCDS). Plug (2004) relies on 610 adoptees in the Wisconsin Longitudinal Survey (WLS) to estimate how parental education impacts child education through the family environment

⁹In terms of empirical approach, we are aware of only one other nurture study that make use of children conceived through sperm and egg donations. Frances Rice et al. (2009) examine how prenatal smoking affects child outcomes using a survey sample of IVF-treated mothers with and without donor children treated in several UK and US fertility clinics. Their main finding is that prenatal smoking reduces birth weight in genetically unrelated and related children. Their approach has two potential problems: the assignment of donors to recipient mothers may not be random, and survey response may be selective. Our study differs from theirs in focus and approach. We are not only asking a different question, but we are also answering it with a more convincing empirical approach and better data. We exploit the quasirandom assignment of egg donors to more credibly identify the mother-child skill relationship, net of genetic skill transfers. We use an administrative sample with information on all donor treated mothers in Denmark.

3 IVF in Denmark: Institutions and Data

In this section, we describe the institutional setting for IVF treatments in Denmark, with emphasis on the use of donor eggs and sperm, and discuss how we construct our data from several administrative registers.

IVF Institutions

Danish couples who experience fertility problems typically visit their general practitioner for medical advice and fertility testing. When childless couples are medically diagnosed as infertile, their general practitioner can refer them to a fertility clinic or hospital. In case the women in infertile couples are below the age of 40, they are entitled to three IVF treatments at no cost.¹⁰ Each year, about 2,500 couples start an IVF-treatment and the average success rate per IVF treatment is 25-30 percent. Most couples undergo 3 to 4 treatments, leading to an overall success rate of 70-75 percent.

The standard IVF procedure works by collecting eggs, fertilizing eggs with sperm in a laboratory environment, and implanting the most promising embryo(s) back into the womb. Most IVF treatments involve the couples' own eggs and sperm. Some IVF treatments, however, involve either donor eggs or donor sperm. Danish law prohibits the fertilization of donor eggs with donor sperm, the argument being that the child should be genetically related to at least one of the parents. This implies that children conceived with donor eggs are genetically unrelated to their mother but genetically related to their father, and vice versa.

Infertile couples may need donors for a number of reasons: some women experience premature menopause, are born without ovaries, or had their ovaries removed; some men experience low sperm counts, or tube blockages; and some women and men have damaged reproductive organs (due to, for instance, previous cancer treatments) or carry possible dangerous genetic diseases. For those couples unable to produce viable eggs or sperm on their own, IVF treatments with donors may be the best alternative to get pregnant and have children.

The process of using donors in Denmark is highly regulated. Over the period

¹⁰While Danish law has set 46 as the maximum treatment age, public clinics generally use 40 as the threshold. The annual costs for IVF-related medication is born by the couple and amounts to about 4,000 DKK annually (which corresponds to US\$640 in 2016). Free IVF treatments applies to first-born children only. In our study, single women were not allowed to undergo IVF treatments. A law change in 2007, however, made it possible for single women to undergo IVF treatment with donor sperm.

we consider, egg and sperm donations were anonymous. Donor recipients (as well as the children born with donated eggs or sperm) were not informed about the identity of their donor, and vice versa, donors were not informed about the identity of their donor recipient. Donors were neither informed about the outcome of the treatment (that is, whether their donated eggs or sperm resulted in a pregnancy) nor could they claim legal parenthood over the children born with their donated eggs or sperm. Parents of donor children are encouraged to tell their children about their donor status around the age of five but are not legally obliged to do so.

Danish law also regulates who can become an egg donor. Over the study period we consider, only women who underwent IVF treatment themselves were allowed to donate eggs. Donor eggs were thus *surplus eggs* from IVF treated women who produced more eggs than needed for their own IVF treatment. To ensure donor egg quality, egg donor candidates were medically screened before they could donate eggs and, once approved, egg donors had to be younger than 35 when they donated eggs. Monetary compensation was not allowed. Only few women volunteer to donate eggs; over the 1995-2007 period, there was a donor egg shortage (donor egg demand exceeded donor egg supply).

Danish law is less restrictive for sperm donors. While the law dictated that, in IVF treatments, only clinics could buy donor sperm from anonymous donors, most men were allowed to donate sperm. Like egg donor candidates, sperm donor candidates were medically screened. Sperm donor candidates with a family history of serious hereditary mental and physical disorders were rejected. Once approved, candidates were repeatedly tested for infectious diseases for the full duration of the donation period. For sperm donations, a small monetary compensation was allowed. Many men donate sperm; over the 1995-2007 period, sperm banks held enough sperm to treat all infertile couples in need of donor sperm (donor sperm supply exceeded donor sperm demand).

For infertile couples treated with donor eggs, IVF clinics follow a three-stage IVF procedure. The first stage involves a hormone medication treatment to prepare the uterus for egg reception. The second stage involves the fertilization of donor eggs with sperm of the recipient's husband or partner. The third and final stage involves the fertilized donor-egg implantation into the recipient's uterus. Given the shortage of donor eggs, couples had to wait several years before they received treatment. In 2008, for instance, the average waiting time was 3.2 years (Larsen et al. 2009). For infertile couples treated with donor sperm, however, IVF clinics buy their donor sperm from sperm banks and just

follow the standard IVF procedure. Given the excess supply of donor sperm, there were no waiting lists for IVF treatments with donor sperm.

IVF Register

In our empirical analyses, we exploit data from the Danish IVF register, currently held by the Danish Health Data Authority (Sundhedsdatastyrelsen).¹¹ The register covers information on all IVF treatments taking place in public and private fertility clinics and hospitals in Denmark. We focus on the period 1994 to 2007. It covers information on the main reason for infertility, the mode of treatment, the use of donor eggs and donor sperm, the number of eggs retrieved from the womb, the number of fertilized eggs transferred back, the date of treatment and clinic identifiers. It also records whether treated women agreed to donate eggs and, if so, how many. We have merged the IVF register to other administrative registers to get longitudinal information on standard demographic variables (including birth year, gender, Danish citizenship, marital status, number of children, and education) running from 1991 to 2016 and standard labor market variables (including labor force status and annual earnings) running from 1991 to 2012.

To study intergenerational mobility patterns in education, we require educational outcomes of IVF-treated parents and their children. We use data from the Danish Education Register, which holds records on educational achievement in primary, lower and upper secondary, and tertiary education from the early 1970s onwards. For parents, we observe realized educational outcomes and take years of schooling as our main parental outcome. For their children, we observe test scores taken from multiple nationwide tests (including 4 tests in reading and 2 tests in math) that were introduced in Danish primary and lower secondary schools in 2010. Appendix Table A gives an overview over the years in which the various tests were taken. Most children in our sample window have taken 3 to 4 tests. Our main child outcome is the average of all available standardized test scores in reading and math.¹²

 $^{^{11}}$ Lundborg, Plug, and Rasmussen (2017) analyze the IVF register in another context: they exploit IVF treatment success at the first IVF treatment as a natural experiment to estimate the causal effect of having children on the career of women.

 $^{^{12}}$ While we can separately examine test score performance in reading and math, we prefer the overall average of multiple test scores as the main child outcome for two reasons. First, and most importantly, it raises precision and reduces the influence of outliers. Second, it accounts for parental influences that possibly spill over across the different tests. We report separate results for math and reading in the sensitivity section.

Table 1Summary statistics

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		non donor children	sperm donor children	egg donor children	all other children	all adopted children	Korean adopted children
		(1)	(2)	(3)	(4)	(5)	(6)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		Child cha	racteristics	and outco	mes:		
male birth year 0.51 2001.86 0.53 2001.41 0.56 2001.8 0.51 2001.05 0.44 2001.65 0.66 2001.65 number of children $19,509$ 820 157 $650,930$ $2,674$ 232 Parent characteristics and outcomes:Pretreatment characteristics years of schooling father college education father (1/0) 0.30 0.29 0.97 0.36 0.25 0.37 0.37 0.36 0.25 Danish citizenship father (1/0) 0.44 0.95 0.97 0.97 0.95 0.96 0.89 0.97 0.97 0.97 0.97 birth year father 13.28 13.28 13.16 12.75 12.98 12.98 13.54 13.54 13.81 0.32 college education mother (1/0) 0.41 0.41 0.39 0.32 0.37 0.48 0.52 Danish citizenship mother (1/0) 0.41 0.41 0.39 0.32 0.37 0.48 0.48 0.52 0.97 Danish citizenship egg donor $-$ $-$ 	standardized test score	0.12	0.26	0.17	0.01	-0.03	0.21
birth year2001.862001.412000.182001.052001.752001.65number of children19,509820157650,9302,674232Parent characteristics and outcomes:Pretreatment characteristicsyears of schooling father12.9012.7113.0412.5913.1613.17college education father (1/0)0.300.290.360.250.370.36Danish citizenship father (1/0)0.950.970.950.890.970.97birth year father13.2813.1612.7512.9813.5413.81college education mother (1/0)0.410.390.320.370.480.52Danish citizenship mother (1/0)0.941968.791964.131970.831967.641964.65years of schooling egg donorDanish citizenship mother (1/0)0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donorcollege education egg donorDanish citizenship egg donor <td< td=""><td>male</td><td>0.51</td><td>0.53</td><td>0.56</td><td>0.51</td><td>0.44</td><td>0.66</td></td<>	male	0.51	0.53	0.56	0.51	0.44	0.66
number of children 19,509 820 157 650,930 2,674 232 Parent characteristics and outcomest Pretreatment characteristics 12,90 12,71 13,04 12,59 13,16 13,17 college education father (1/0) 0.30 0.29 0.36 0.25 0.37 0.36 Danish citizenship father (1/0) 0.95 0.97 0.95 0.89 0.97 0.97 birth year father 1966.43 1964.87 1962.82 1965.99 1963.63 gears of schooling mother 13.28 13.16 12.75 12.98 13.54 13.64 college education mother (1/0) 0.41 0.39 0.32 0.37 0.48 0.52 pars of schooling egg donor - - 12.88 - - - college education egg donor - - 0.97 - - - pars of schooling egg donor (1/0) - - 0.36 - - - parental leave day	birth year	2001.86	2001.41	2000.18	2001.05	2001.75	2001.65
Parent characteristics and outcomes:Pretreatment characteristicsyears of schooling father (1/0) 12.90 12.71 13.04 12.59 13.16 13.17 college education father (1/0) 0.30 0.29 0.36 0.25 0.37 0.36 Danish citizenship father (1/0) 0.55 0.97 0.95 0.89 0.97 0.97 birth year father 1966.43 1964.87 1962.82 1968.25 1965.99 1963.63 years of schooling mother college education mother (1/0) 0.41 0.39 0.32 0.37 0.48 0.52 Danish citizenship mother (1/0) 0.94 0.95 0.95 0.89 0.97 0.99 birth year mother 1968.94 1968.79 1964.13 1970.83 1967.64 1964.65 years of schooling egg donor $ 0.32$ $ -$ college education egg donor $ 0.97$ $ -$ panish citizenship egg donor (1/0) $ 0.96$ $ -$ birth year egg donor $ 0.36$ $ -$ first 5 years following child birth/ change in annual earnings father 0.08 -0.00 0.10 $ -$ parental leave days father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mo	number of children	19,509	820	157	650,930	$2,\!674$	232
Pretreatment characteristicsyears of schooling father12.9012.7113.0412.5913.1613.17college education father (1/0)0.300.290.360.250.370.36Danish citizenship father (1/0)0.950.970.950.890.970.97birth year father1966.431964.871962.821968.251965.991963.63years of schooling mother13.2813.1612.7512.9813.5413.81college education mother (1/0)0.410.390.320.370.480.52Danish citizenship mother (1/0)0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donorcollege education egg donor0.32college education egg donor0.32college education egg donor0.32parental eave days father19.2216.7712.1216.5227.68-college in annual earnings father0.08-0.000.10parental leave days mother333.78298.46297.65272.33254.22-divorce0.170.210.110.270.34-change in annual earnings mother19,50		Parent ch	aracteristic	s and outc	omes:		
years of schooling father12.9012.7113.0412.5913.1613.17college education father $(1/0)$ 0.300.290.360.250.370.36Danish citizenship father $(1/0)$ 0.950.970.950.890.970.97birth year father1966.431964.871962.821968.251965.991963.63years of schooling mother13.2813.1612.7512.9813.5413.81college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donorcollege education egg donor (1/0)college education egg donor0.32college donor0.97college donor0.97college donor0.97college donor0.97college donor0.97college donor0.97missing information egg donor1968.30	Pretreatment characteristics						
college education father $(1/0)$ 0.300.290.360.250.370.36Danish citizenship father $(1/0)$ 0.950.970.950.890.970.97birth year father1966.431964.871962.821968.251965.991963.63years of schooling mother13.2813.1612.7512.9813.5413.81college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor0.32college education egg donor0.97college education egg donor0.97Danish citizenship egg donor0.97Danish citizenship egg donor0.97birth year egg donor0.97missing information egg donor0.97first 5 years following child birth)0.16change in annual earnings father0.08-0.000.10parental leave days mother333.78298.46297.65272.33254.22-	years of schooling father	12.90	12.71	13.04	12.59	13.16	13.17
Danish citizenship father $(1/0)$ 0.95 0.97 0.95 0.89 0.97 0.97 birth year father1966.431964.871962.821968.251965.991963.63years of schooling mother13.2813.1612.7512.9813.5413.81college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor0.32college education egg donor0.32college education egg donor0.32panish citizenship egg donor (1/0)0.97birth year egg donor1968.30missing information egg donor0.36Posttreatment outcomes(first 5 years following child birth) change in annual earnings father19.08-0.000.10change in annual earnings mother-0.10-0.17-0.09querce days mother333.78298.46297.65272.33254.22-divorce0.170.210.110.270.34-number of children19,509	college education father $(1/0)$	0.30	0.29	0.36	0.25	0.37	0.36
birth year father1966.431964.871962.821968.251965.991963.63years of schooling mother13.2813.1612.7512.9813.5413.81college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor0.32college education egg donor0.97Danish citizenship egg donor (1/0)0.97Danish citizenship egg donor1968.30Danish citizenship egg donor0.960.36Danish citizenship egg donor0.97Danish citizenship egg donor0.97missing information egg donor0.36Posttreatment outcomes(first 5 years following child birth) change in annual earnings father0.08-0.000.10parental leave days mother-0.10-0.17-0.09divorce0.170.210.110.270.34-mumber of children19,	Danish citizenship father $(1/0)$	0.95	0.97	0.95	0.89	0.97	0.97
years of schooling mother13.2813.1612.7512.9813.5413.81college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor0.32college education egg donor0.32Danish citizenship egg donor (1/0)0.97Danish citizenship egg donor1968.30birth year egg donor1968.30missing information egg donor0.36Posttreatment outcomes(first 5 years following child birth)0.000.10parental leave days father19.2216.7712.1216.5227.68change in annual earnings mother-0.10-0.17-0.09parental leave days mother333.78298.46297.65272.33254.22divorce0.170.210.110.270.34number of children19,509820157650,9302,674232196	birth year father	1966.43	1964.87	1962.82	1968.25	1965.99	1963.63
college education mother $(1/0)$ 0.410.390.320.370.480.52Danish citizenship mother $(1/0)$ 0.940.950.950.890.970.99birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor12.88college education egg donor0.32Danish citizenship egg donor0.32Danish citizenship egg donor0.97Danish citizenship egg donor1968.30Danish citizenship egg donor1968.30missing information egg donor0.36Posttreatment outcomes(first 5 years following child birth)-0.08-0.000.10change in annual earnings father19.2216.7712.1216.5227.68-change in annual earnings mother-0.10-0.17-0.09parental leave days mother333.78298.46297.65272.33254.22-divorce0.170.210.110.270.34-number of children19,509820157650,9302,674232number of children19,509820157650,9302,674232	years of schooling mother	13.28	13.16	12.75	12.98	13.54	13.81
Danish citizenship mother $(1/0)$ 0.94 0.95 0.95 0.89 0.97 0.99 birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor $ 12.88$ $ -$ college education egg donor $ 0.32$ $ -$ Danish citizenship egg donor (1/0) $ 0.97$ $ -$ birth year egg donor $ 1968.30$ $ -$ missing information egg donor $ 0.36$ $ -$ Posttreatment outcomes(first 5 years following child birth) $ 0.08$ -0.00 0.10 $ -$ change in annual earnings father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of children $19,509$ 820 157 $650,930$ $2,674$ 232	college education mother $(1/0)$	0.41	0.39	0.32	0.37	0.48	0.52
birth year mother1968.941968.791964.131970.831967.641964.65years of schooling egg donor $ -$ Danish citizenship egg donor $ 0.32$ $ -$ Danish citizenship egg donor $ 0.97$ $ -$ birth year egg donor $ 1968.30$ $ -$ missing information egg donor $ 0.36$ $ -$ Posttreatment outcomes(first 5 years following child birth) $ 0.08$ -0.00 0.10 $ -$ parental leave days father19.2216.7712.1216.5227.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46297.65272.33254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of children $19,509$ 820 157 $650,930$ $2,674$ 232	Danish citizenship mother $(1/0)$	0.94	0.95	0.95	0.89	0.97	0.99
years of schooling egg donor - - 12.88 - - - college education egg donor - - 0.32 - - - Danish citizenship egg donor - - 0.97 - - - Danish citizenship egg donor - - 0.97 - - - birth year egg donor - - 1968.30 - - - missing information egg donor - - 0.36 - - - Posttreatment outcomes - - 0.36 - - - - first 5 years following child birth) - - - - - - change in annual earnings father 19.22 16.77 12.12 16.52 27.68 - change in annual earnings mother -0.10 -0.17 -0.09 - - - parental leave days mother 333.78 298.46 297.65 272.33 254.22 - divorce 0.17 0.21 0.11 0.27 <	birth year mother	1968.94	1968.79	1964.13	1970.83	1967.64	1964.65
college education egg donor - - 0.32 - - - Danish citizenship egg donor - - 0.97 - - - birth year egg donor - - 1968.30 - - - missing information egg donor - - 0.36 - - - Posttreatment outcomes - - 0.36 - - - - (first 5 years following child birth) - - - 0.10 - - - - parental leave days father 19.22 16.77 12.12 16.52 27.68 - - change in annual earnings mother -0.10 -0.17 -0.09 - - - parental leave days mother 333.78 298.46 297.65 272.33 254.22 - divorce 0.17 0.21 0.11 0.27 0.34 - number of children 19,509 820 157 650,930 2,674 232 number of children 19,509 <	years of schooling egg donor	_	_	12.88	_	_	_
Danish citizenship egg donor $(1/0)$ - - 0.97 - - - birth year egg donor - - 1968.30 - - - - missing information egg donor - - - 0.36 - - - Posttreatment outcomes (first 5 years following child birth) - - - - - change in annual earnings father 0.08 -0.00 0.10 - - - - parental leave days father 19.22 16.77 12.12 16.52 27.68 - change in annual earnings mother -0.10 -0.17 -0.09 - - - parental leave days mother 333.78 298.46 297.65 272.33 254.22 - divorce 0.17 0.21 0.11 0.27 0.34 - number of children 19,509 820 157 650,930 2,674 232 number of children 19,509 617 127 406 109 2,328 196	college education egg donor	_	_	0.32	_	_	_
birth year egg donor - - 1968.30 -	Danish citizenship egg donor $(1/0)$	_	_	0.97	_	_	_
missing information egg donor $ 0.36$ $ -$ Posttreatment outcomes (first 5 years following child birth) $ 0.00$ 0.10 $ -$ parental leave days father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children 19,509 820 157 $650,930$ $2,674$ 232 number of children 19,509 820 157 $650,930$ $2,674$ 232	birth year egg donor	_	_	1968.30	_	_	-
Posttreatment outcomes (first 5 years following child birth) change in annual earnings father 0.08 -0.00 0.10 $ -$ parental leave days father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of children $14,200$ 617 127 $406,100$ $2,232$ 196	missing information egg donor	_	_	0.36	-	_	_
(first 5 years following child birth) 0.08 -0.00 0.10 $ -$ change in annual earnings father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children 19,509 820 157 650,930 2,674 232 number of mothers 14,200 617 127 406109 $2,238$ 196	Posttreatment outcomes						
change in annual earnings father 0.08 -0.00 0.10 $ -$ parental leave days father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of children $14,200$ 617 127 $406,100$ $2,232$ 196	(first 5 years following child birth)						
parental leave days father 19.22 16.77 12.12 16.52 27.68 $-$ change in annual earnings mother -0.10 -0.17 -0.09 $ -$ parental leave days mother 333.78 298.46 297.65 272.33 254.22 $-$ divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of children $14,200$ 617 127 $406,100$ $2,232$ 196	change in annual earnings father	0.08	-0.00	0.10	_		-
change in annual earnings mother parental leave days mother -0.10 333.78 -0.17 298.46 -0.09 297.65 $-$ 272.33 $-$ 254.22 divorce 0.17 0.21 0.11 0.27 0.34 $-$ number of children number of mothers $19,509$ 820 157 $650,930$ $2,674$ 232 14,200 617 127 $406,100$ $2,228$ 196	parental leave days father	19.22	16.77	12.12	16.52	27.68	_
parental leave days mother 333.78 298.46 297.65 272.33 254.22 - divorce 0.17 0.21 0.11 0.27 0.34 - number of children 19,509 820 157 650,930 2,674 232 number of mothers 14,200 617 127 406,100 2,228 196	change in annual earnings mother	-0.10	-0.17	-0.09	_	_	_
divorce 0.17 0.21 0.11 0.27 0.34 - number of children 19,509 820 157 650,930 2,674 232 number of mothers 14,200 617 127 406,100 2,222 196	parental leave days mother	333.78	298.46	297.65	272.33	254.22	_
number of children $19,509$ 820 157 $650,930$ $2,674$ 232 number of mothers $14,200$ 617 127 $406,100$ $2,222$ 106	divorce	0.17	0.21	0.11	0.27	0.34	-
number of mothers 14200 617 127 406100 2222 106	number of children	19,509	820	157	650,930	2,674	232
14,200 017 127 400,109 2,328 190	number of mothers	14,200	617	127	406,109	2,328	196

Note–The table shows descriptive statistics for different intergenerational samples of children with test scores: (i) all nondonor IVF children; (ii) all sperm donor IVF children; (iii) all egg donor IVF children; (iv) a representative sample of all other children; (v) a representative sample of adopted children; and (vi) a representative sample of adopted children from South Korea. Appendix Table B contains the definition of all variables

Our main analysis sample is restricted to those IVF children for whom we observe the score of at least one nationwide test. We select 19,509 children in the IVF register, who were not conceived through either donor sperm or donor eggs, 820 sperm donor children, and 157 egg donor children. For the egg donor sample, we are able to match (with some certainty) a sizable fraction of children to their egg donors; that is, we have information on egg donors for 97 children. See Section 4 for details.

For comparison purposes, we construct two additional samples. The first sample is the representative sample, which is a 30 percent random sample of families with nonadopted children born in similar years as the IVF children. The second sample is our adoption sample, which is a 30 percent random sample of families with children adopted from abroad. We observe 650,930 nonadopted and non-IVF children and 2,674 adopted children. Of these adoptees, 232 were adopted from South-Korea.¹³

Table 1 provides sample means for the intergenerational samples. We make three informative comparisons. First, IVF children, and donor children in particular, perform much better in nationwide tests than children in the representative sample. Compared to these representative children, for instance, we find that IVF children have 0.11-0.25 standard deviation higher test scores. Second, IVF treated parents and representative parents tend to be different. IVF treated parents are older than the representative parents. IVF treated parents are in more stable relationships than the representative parents. When we look at education, which is the parental characteristic at the center of this study, the differences are less pronounced. IVF treated fathers (not mothers) are better educated, but not by much, than the representative fathers. When compared to the education levels of parents in the adoption samples, however, IVF treated parents and representative parents appear much more similar than different. And third, we find that, among IVF treatments with egg donors, donor recipients are much older than egg donors, which is consistent with the age restriction of 35 imposed on egg donors. But when it comes to educational attainment, treated women and their donors are nearly identical.

 $^{^{13}}$ Previous adoption studies with Korean-born adoptees can approximate quasi-random assignment of children to families by exploiting a first-come first-serve principle in the adoption application process (Sacerdote 2007, Hægeland et al. 2010, and Fagereng, Mogstadt, and Rønning 2018). For comparison purposes, we also present results for Korean-born adoptees for whom the first-come first-serve principle applies. We need to assume random assignment, though, because we do not know the application position of prospective adoptive parents.

4 Assignment of Donors to Families

In this section we document how the sperm and eggs of donors were assigned to Danish couples in IVF treatments. We are particularly interested in the extent to which donor assignment occurs randomly.

The Donor Assignment Process

As we mentioned above, donor assignment is bound by strict rules on donor anonymity. While donors are strictly anonymous, prospective parents can state their preferences for donors on five dimensions: skin color, hair color, eye color, weight and length. These donor preferences are expressed in a donor market with an excess supply of sperm donors and excess demand for egg donors, which implies that donor preferences are likely met for sperm donors, but not for egg donors.

With an excess supply of sperm donors, sperm donors are assigned based on the prospective parents' preferences, which makes donor assignment in principle random conditional on these stated preferences. The IVF register does not record these preferences. In any case, our intergenerational results for fathers are not consistent with any strong selection based on donor preferences, as we will see later.

With an excess demand for egg donors, prospective parents in need of an egg donor are placed on a waiting list. Fertility clinics organize their own waiting lists. There are in total 21 fertility clinics (including hospitals that offer IVF treatments). Prospective parents choose one fertility clinic, which together with the shortage of eggs means that donor assignment depends on the position on the clinic-specific waiting list rather than donor preferences. To substantiate this claim, we quote from the guidelines of one of the largest IVF clinics in Denmark (Ciconia): "Because of the long waiting time, it is not possible to match physical characteristics. You are offered donor eggs in the same order as you have been put on the waiting list." (Ciconia 2015). The IVF register contains detailed records of the date and place of the IVF treatment. If we take the date of the first donor treatment (measured in calendar months) and a full set of clinic indicators to accurately proxy the clinic-specific waiting list order, the assignment of donor eggs to prospective parents should be as good as random conditional on the calendar month of first donor treatment and clinic fixed effects.

Is Donor Assignment Conditionally Random (for Women)?

With the IVF register at hand, we can identify (with some certainty) the recipients' egg donor and empirically assess whether egg donor assignment is conditionally random. In our IVF context where the nurturing influence of parental education is the treatment of interest, the natural verification test is to link the educational attainment of donor recipients to the educational attainment (or any other pre-assignment characteristic) of their donors. Conditional random assignment would predict zero associations, that is, after taking account of the assignment control variables.

The IVF register contains information on women who provide donor eggs and women who receive fertilized donor egg implants. For egg donors, the register records the number of donor eggs extracted and the extraction date. For donor egg recipients, the register records the number of donor eggs implanted and the treatment date, which may represent either the egg-implantation date (which occurs after the donors' extraction date) or the preparation-for-implantation date (which occurs before the donors' extraction date). With the treatment histories, we can link donor egg recipients to their egg donors based on the (correct) premise that fertility clinics take their egg donors from the women they treat and predominantly use fresh (fertilized) donor eggs as egg implants.¹⁴ We define a match (with some certainty) if donor egg recipients and egg donors are treated at the same clinic and the recipients' treatment date occurs within one week after the donors' extraction date (when recipients receive the embryo implants) or within seven weeks before the donors' extraction date (when recipients receive medication to prepare their uterus for pregnancy and egg implants). The matched pairs constitute our verification sample. Over the 1994-2007 period, we are able to identify 533 matches treated at the same clinic at the same period. With matched-pair observations, egg donors and egg donor recipients may enter the sample several times. If egg donors produce enough viable donor eggs, they can serve multiple donor recipients. If previous IVF attempts with donor eggs failed, egg donor recipients may be treated with eggs from multiple donors (in multiple treatments). There are, in total, 346 different donor recipients and 419 different egg donors.¹⁵

 $^{^{14}}$ Over the period we study, most treatments involve fresh embryo transfers. For the years 1994-1995, Westergaard et al. (2000) report that 90 percent of all embryo transfers were fresh embryo transfers. For the years 2006-2011, when we have information on fresh and frozen embryo transfers in IVF treatments, 86 percent of egg donor treatments made use of fresh (fertilized) eggs.

¹⁵The egg donor recipients in the verification sample only partially overlap with those

	dr acter 15	ucs and uc			arion		
	yes	ars of	don	or	юр	nor	
	edu	cation	ag	е	citize	nship	
	(1)	(2)	(3)	(4)	(5)	(9)	
recipient years of education	-0.002	-0.004	-0.012	0.014	-0.002	-0.002	
5	0.031	0.029	0.058	0.058	0.002	0.002	
recipient birth year		0.029		-0.015		-0.0001	
		0.013^{**}		0.032		0.001	
recipient citizenship		0.116		0.045		0.048	
		0.264		0.370		0.033	
R-squared	0.008	0.020	0.005	0.007	0.003	0.011	
F-test (joint significance)		1.93		0.10		0.93	
number of observations	533	533	533	526	533	533	
number of egg donors	419	419	419	419	419	419	
number of donor recipients	364	364	364	364	364	364	
clinic FE and treatment date controls	>	>	>	>	>	>	

 Table 2

 Relationship between egg donor characteristics and donor recipient educatic

Note-The dependent variables are egg donor years of education, age and Danish citizenship. In columns 1, 3 and 5, we regress donor characteristics on recipient years of education. In columns 2, 4 and 6, we regress donor characteristics on recipient years of education, birth year and Danish citizenship. All regressions control for the first donor treatment date (in calendar months) and a full set of clinic indicators. The F-test reports whether the recipient characteristics are jointly significant (low values indicate statistically insignificance). Standard errors are clustered by recipients and shown in italics; * indicates significance at 10 percent level, ** indicates significance at 5 percent level, and *** at 1 percent level.

Table 2 presents the test results. In particular, we estimate how the preassignment characteristics of egg donors (education, age, and Danish citizenship) relate to the educational attainment of egg donor recipients after controlling for the waiting list variables (month of first donor treatment and clinic fixed effects). In columns 1, 3 and 5, we regress the three donor outcomes on recipient years of education and waiting list controls. In columns 2, 4 and 6, we augment the regression with recipient birth year and citizenship. We find that, conditional on the recipient's position on the waiting list, there is no relationship between the three donor characteristics and the recipient's education. All the education estimates are statistically insignificant and close to zero. We also find that there is no relationship between the three donor characteristics and almost all other recipient characteristics. Test results are, in general, as one would expect with conditional random assignment of donor eggs to donor recipients. One anomaly is the association between donor education and recipient birth year, which is small but statistically significant. We attribute this to chance and take comfort from the facts that, first, the three recipient characteristics together are not statistically significant (the F-statistic of 1.93 comes with a p-value of 0.33), second, the zero estimates for recipient education (which is the key treatment variable in our analysis) are insensitive to the inclusion of other recipient characteristics and, third, the intergenerational regressions that follow always control for recipient birth year.

5 Empirical Strategy

In our main analysis, we take the sample of IVF treated families and distinguish parents of nondonor children from parents of donor children. We begin by presenting a simple reduced-form intergenerational mobility model where both parents influence the educational achievement of their nondonor children

$$Y_i^c = \beta^c + \beta^m Y_i^m + \beta^f Y_i^f + \gamma' X_i + e_i^c.$$

$$\tag{1}$$

In this regression model Y_i^c represents an intermediate educational achievement outcome (measured in nationwide achievement tests in primary and lower sec-

in the intergenerational sample. The verification sample, for instance, contains more eggdonor recipients because we sampled all egg donor recipients including those whose donor egg implants did not result in children and those whose children had no available test scores. Of the 533 donor treatments in the verification sample, only 118 were successful and lead to children. We observe test scores for 97 children. The intergenerational sample, on the other hand, contains more egg-donor recipients with children because we could not match all egg-donor recipients to their donors.

ondary education) of child c born and raised in family i with mother m and father f, Y_i^m and Y_i^f represent the educational achievement of the child's mother and father (measured in completed years of schooling), X_i represents a set of pre-determined child, family, and treatment variables (including the child's gender and birth year, the mother's and father's birth year and citizenship, the first treatment date and a full set of clinic indicators), and e_i^c represents exogenous child-specific characteristics. We measure birth year in continuous years and first treatment date in continuous calendar months (because of sample size considerations). The intergenerational coefficients β^f and β^m measure the intergenerational association between the educational achievement of genetically related children and parents and represent an unknown blend of nurture and nature influences. With samples of nondonor children, ordinary least squares (OLS) estimation of (1) yields estimates of β^f and β^m .

Our data on donor children allow us to isolate the nurturing component from the intergenerational coefficients β^m and β^f . To explain how we identify the nurturing effects, we introduce a simple hypothetical intergenerational transmission model, akin to the transmission model of Björklund, Lindahl, and Plug (2006), where all four parents can influence the child's education

$$Y_{ijk}^{c} = \alpha^{c} + \alpha_{1}^{m} Y_{j}^{m} + \alpha_{2}^{m} Y_{i}^{m} + \alpha_{1}^{f} Y_{k}^{f} + \alpha_{2}^{f} Y_{i}^{f} + \theta' X_{i} + e_{ijk}^{c}, \qquad (2)$$

where the subscripts c, i, j and k stand for child c raised in family i is conceived by egg donor j and sperm donor k, the superscripts m and f stand for the child's mother and father, Y_j^m and Y_k^f represent the measures of the educational achievement of the child's genetically related donor mothers and fathers, Y_i^m and Y_i^f represent the measures of the educational achievement of the child's genetically unrelated mothers and fathers, X_i represents a set of pre-determined child, family, and treatment variables (as defined in (1)), and e_{ijk}^c represents unobserved child-specific characteristics.¹⁶

The intergenerational coefficients α_1^f and α_1^m measure the intergenerational associations between the educational achievement of the child and donors and represent the nature effects. The intergenerational coefficients α_2^f and α_2^m measure the intergenerational associations between the educational achievement of the child and genetically unrelated parents and represent the nurture effects.

¹⁶While this is clearly a very simplified description of the real world (it ignores, for instance, that the test scores of children can be affected by possible interactions between nature and nurture, which we discuss later in this section), Björklund, Lindahl, and Plug (2006) show that this very simple model characterizes the intergenerational transmission of education surprisingly well, at least for Swedes born in 1962-1966.

These nurture effects, which are the prime targets of estimation, must be interpreted broadly and capture the causal influence of parental education and any other unobserved parenting skill that is correlated with it.¹⁷

Before turning to the identification of nurture effects α_2^f and α_2^m using samples of donor children, we first reflect on the information we have on the parents. We have complete information on the rearing mothers and fathers. We have incomplete information on the egg donors (that is, we are able to match 64 percent of the children to their egg donors). We have no information on sperm donors. Our intergenerational analysis therefore centers around the rearing mothers and fathers and treats egg and sperm donors as omitted variables (unless indicated otherwise). This will cause no problem for identifying the nurture effect for mothers, but may complicate the identification of the nurture effect for fathers.

Identifying Nurture Effects (Internal Validity)

The identification of nurture effects α_2^f and α_2^m on samples of donor children without (using) information on donors must assume that donor assignment in IVF treatments is either random or related to variables that we observe and control for. The argument is simple: under (conditional) random donor assignment, we know that the characteristics of rearing parents and their donors are independent and that our nurture effect estimates are not affected once we exclude (or include) donor characteristics from (in) the intergenerational mobility regressions.

In case of egg donor children raised in family *i* with genetically unrelated mother *i*, egg donor *j*, and genetically related father *i* (with identical Y_i^f and Y_k^f), donor assignment follows a clinic-specific waiting list principle. With controls for first treatment date (measured in calendar months) and a full set of clinic indicators to account for the position mothers take on the waiting list, we can ignore the influence of the donor's genes, as if α_1^m is zero, and rewrite the intergenerational transmission model as

$$Y_{ij}^{c} = \alpha^{c} + \alpha_{2}^{m} Y_{i}^{m} + (\alpha_{1}^{f} + \alpha_{2}^{f}) Y_{i}^{f} + \theta' X_{i} + e_{ij}^{c}.$$
(3)

¹⁷The intergenerational mobility models for nondonor and donor children are connected in a relatively straightforward way. For nondonor children born and raised in family *i*, we know that Y_j^m and Y_k^f are identical to Y_i^m and Y_i^f . If we assume that nurture and nature effects are similar in families with nondonor and donor children, model (2) should collapse to (1) where the intergenerational transmission coefficients $\alpha_1^m + \alpha_2^m$ and $\alpha_1^f + \alpha_2^f$ are identical to β^m and β^f .

With samples of egg donor children, direct estimation of (3) gives us an unbiased nurture estimate of α_2^m .¹⁸

In case of sperm donor children raised in family i with genetically related mother i (with identical Y_i^m and Y_j^m), genetically unrelated father i, and sperm donor k (with unknown Y_k^f), donor assignment is guided by stated donor preferences. Adding these variables to X in the intergenerational transmission model would analogously eliminate the genetic influences of the sperm donor

$$Y_{ik}^{c} = \alpha^{c} + (\alpha_{1}^{m} + \alpha_{2}^{m})Y_{i}^{m} + \alpha_{2}^{f}Y_{i}^{f} + \theta'X_{i} + e_{ik}^{c}.$$
(4)

The problem is that we do not observe these preference variables. With donor preferences excluded from (4), the estimate of α_2^f may be biased and capture not only the nurture effect of the rearing father but also part of the nature effect of the donor, that is, if the educational outcomes of rearing fathers and sperm donors are somehow related through the omitted preferences. There are, however, good a priori reasons to believe that these omitted preferences cause only little (upward) bias, given that donors are anonymous and parents can only choose out of five donor traits that are at best crude proxies for the donor's educational attainment.

Equations (3) and (4) relate the educational attainment of children to the educational attainment of *both* rearing parents. We purposefully do so to take account of confounding assortative mating effects. Would we exclude the genetically related father from (3), for example, the estimate of α_2^m captures not only the direct influence of the mother's education (representing the nurture effect) but also the indirect influence of the father's education (representing a mixture of nurture and nature effects) because better educated mothers tend to marry better educated fathers. With the educational attainment of both rearing parents in the same specification, we can better separate out the direct effects of the genetically unrelated parent from the indirect effects of the other genetically related parent. In our sensitivity analysis, we provide some evidence on the size of confounding assortative mating effects.

¹⁸Our data on matched egg donors make is possible to estimate the same intergenerational models with the educational attainment of the matched egg donor as additional regressor. If assignment is indeed conditionally random, the estimated nurture effect should not change when we include the matched egg donor's educational attainment.

Extrapolating Nurture Effects (External Validity)

The extrapolation of nurture effects α_2^m and α_2^f obtained with donor children to a larger population of nondonor children must assume that nurture effects are similar in donor and nondonor families.

In our sample of IVF children, we can think of two reasons why the nurture similarity assumption gets violated. The first reason is that nurture effects are heterogeneous, that is, IVF parents with donor children are inherently different from IVF parents with nondonor children in how parents create environments for children to grow up, or in how children respond to their parents' upbringing. The second and related reason is that parents invest differently in donor and nondonor children. We propose two tests for such violations.

One test focuses on heterogenous nurture effects and compares intergenerational skill transfers of genetically related parents in donor and nondonor families. Because one parent in donor families needs to be the biological one, we can estimate the combined nature-nurture effect for at least one of the parents in donor families. If the combined nurture-nature effects are heterogeneous and different in donor and nondonor families, we should find different intergenerational transmission coefficients.

The other test focuses on treatment differentials in terms of time parents spend with children and compares parental leave, labor supply, and divorce decisions of parents in donor and nondonor families. Parents may treat their genetically unrelated children differently for different reasons. If, on the one hand, genetically unrelated parents are parents that miss some evolutionary drive (as suggested by Dawkins (1976, 2006) and partly supported by Case, Lin, and McLanahan (2000, 2001)), they may feel less attached to their donor children, and as a result take up less parental leave, work longer hours (assuming that child rearing is most time intensive when children are young), and face higher divorce risks. If, on the other hand, genetically unrelated parents are parents with a stronger demand for children, they may also feel more attached to their donor children and as a result treat their donor children more favorably, with reversed parental leave, labor supply, and divorce responses.

Exploring Nature-Nurture Interactions

In case nurture and nature interact, the nurture effects we estimate also turn heterogenous and must be interpreted as nurture effect averages, which vary with the innate skills of donor children. A number of theoretical models formalize these nature-nurture interactions. There are nurture-orientated interaction models where the skills produced in early childhood raise the productivity of parental environments and investments in later childhood years (Cunha and Heckman 2007). There are also nature-orientated interaction models where the genes exposed to different parental environments can lead to different gene expressions affecting the skill formation of children differently (Rutter 2006). A rising number of studies empirically test for these nature-nurture interactions. Those studies that estimate behavioral genetic models with interactions using twin siblings reared apart report only few statistically significant interactions (see Plomin, DeFries, and Fulker (1988) for a review on the topic). Those studies that estimate intergenerational associations using adoption data with information on the children's biological and adoptive parents generally find small and mostly insignificant interactions (Björklund, Lindahl, and Plug 2006, Brandén, Lindahl, and Öckert 2017, Black et al. 2019). While these nature-nurture interactions have been given much attention in the literature, valid interaction tests are scarce and often unsupportive.

In our setup, we can only implicitly test for nature-nurture interactions using information on donor children. With a sample of donor children who are genetically related to only one of their parents, we can modify the intergenerational models (3) and (4) and add interactions between educational attainment of the genetically unrelated and related parents to account for possible interactions between the nurture and nature effects

$$Y_{ij}^{c} = \alpha^{c} + \alpha_{2}^{m} Y_{i}^{m} + (\alpha_{1}^{f} + \alpha_{2}^{f}) Y_{i}^{f} + \alpha_{3}^{fm} Y_{i}^{m} Y_{i}^{f} + \theta' X_{i} + e_{ij}^{c},$$
(5)

and

$$Y_{ik}^{c} = \alpha^{c} + (\alpha_{1}^{m} + \alpha_{2}^{m})Y_{i}^{m} + \alpha_{2}^{f}Y_{i}^{f} + \alpha_{3}^{mf}Y_{i}^{m}Y_{i}^{f} + \theta'X_{i} + e_{ik}^{c}.$$
 (6)

Here the coefficients α_3^{fm} and α_3^{mf} serve as implicit interaction tests. In case of positive interaction estimates, the α_3^{fm} and α_3^{mf} coefficients may capture various channels, including interaction effects, child-rearing complementarities between mothers and fathers, and other concavities in the child-parent relationship. In case of zero interaction estimates, however, we consider these channels, including nurture-nature interactions, unlikely.

6 Results

Table 3 presents the main intergenerational transmission estimates for education. We report estimates of regressions of standardized test scores on the years of education of both parents, with controls for the child's gender and birth year, the parents' citizenship and birth year, and where indicated the calendar month of first IVF treatment and clinic fixed effects. These control estimates are not reported. We run separate regressions on samples of IVF treated parents with nondonor, sperm donor, and egg donor children. For the sample of egg donor children, we always control for treatment date and clinic fixed effects to ensure that the assignment of donor eggs is conditionally random.

In columns 1 and 2 we begin with the intergenerational mobility associations obtained from the sample of nondonor children. For both parents, we find that the estimated associations between parental schooling and child test scores are large, positive, and statistically significant indicating that higher educated parents have on average children with higher test scores. The estimated associations do not change when we control for treatment date and clinic fixed effects. The overall magnitudes of these associations, which differ only a little for mothers and fathers, tell us that four more years of parental education of either parent are associated with children having about 0.26-0.30 standard deviation higher test scores. These estimates are comparable to those estimated in previous intergenerational mobility studies (Hægeland et al. 2010).

In columns 3 and 4 we run the same intergenerational mobility regressions on the sample of sperm donor children. Here the intergenerational associations are supposed to take account of the father's genes. For fathers of sperm donor children, we find that the estimates get much smaller. In fact, the estimated associations, which we interpret as nurture effects, are practically zero, statistically insignificant, and (statistically) significantly smaller than those obtained for fathers of nondonor children.¹⁹ For mothers of sperm donor children, however, we find that the intergenerational associations between the educational attainment of mothers and child test scores remain large, positive, and statistically significant, and statistically similar to those obtained for mothers of nondonor children. As before, the estimated associations do not change much

¹⁹A critical concern for our nurture effect interpretation is that the assumption of almost random sperm donations does not hold in regressions without donor preference controls. Would the assignment of sperm donors be selective, part of what we interpret as the nurture effect may in fact be genetic. With nurture estimates close to zero, however, we consider such a bias unlikely. We therefore interpret the nurture effect estimates (as presented) in a causal way.

)					
	nor don childr	or en	spei dor child	rm Jor Iren	egg don childi	s or ten	
	(1)	(2)	(3)	(4)	(5)	(9)	
years of education mother	0.076	0.076	0.061	0.067	0.072	0.073	
years of education father	0.003^{***} 0.064	0.004^{***}	0.016^{***} -0.010	0.018^{***} -0.008	0.024^{***}	0.022^{***}	
years of education donor	0.003***	0.003***	0.014	0.015	0.022***	0.022*** 0.008 0.040	
R-squared	0.129	0.132	0.047	0.085	0.258	0.260	
number of observations	19,509	19,509	820	820	157	157	
clinic FE and treatment date controls		>		>	>	>	

Regressions of child test scores on parent education using IVF children

Table 3

educational attainment measured in the nominal years spent in school. All specifications control for the gender and birth year of children, Danish citizenship and birth year of rearing mothers and fathers, and where indicated the first IVF treatment date (in calendar months) and a full set of clinic indicators. The specifications for mothers of egg donor children always control for IVF clinic fixed effects and IVF treatment date to ensure that donor assignment is conditionally random. The last specification (column of the donors treated in the same year and same clinic. Standard errors are clustered by mothers and shown in italics; * indicates significance at 10 percent level, ** indicates significance at 5 percent level, and *** at 1 percent level. 6) adds a control for educational attainment of the egg donor and an imputation indicator. In case the egg donor is not identified (60 out of 157 observations), we have imputed the educational attainment of the egg donor with the average educational attainment Note-The dependent variable is the averaged standardized achievement test score. The independent variables are the parents'

when we control for treatment date and clinic fixed effects.

In columns 5 to 6 we switch to the sample of egg donor children and report intergenerational associations that take account of the mother's genes. We consider two specifications: without and with controlling for the egg donor's educational attainment. For the rearing mothers of egg donor children, we find that the intergenerational associations are large, positive, and statistically significant, and as large as those obtained for mothers of nondonor children. For fathers of egg donor children, we also find positive, sizable, and statistically significant associations that are practically identical to those obtained for fathers of nondonor children. When we control for the donor's educational attainment (measured in years of schooling), we expect the same intergenerational association for rearing mothers (because the assignment of donor eggs is essentially random) and a zero intergenerational association for the matched donors (because the estimated associations for mothers of nondonor children and rearing mothers of egg donor children are practically the same). This is indeed what we find. The intergenerational associations for the rearing parents hardly change. The intergenerational association for the matched donors is small, although imprecisely estimated.²⁰

With small donor samples, there is the concern of imprecise nurture effect estimates. We can draw, nonetheless, three tentative conclusions from these findings. The first one is that the education of genetically unrelated fathers is of little help to their children's test scores; that is, the estimated zero nurturing effect of paternal education (identified on a sample of 820 sperm donor children) is precise enough to statistically rule out effect sizes larger than 0.021 (which is about one third of the overall intergenerational association we estimate for nondonor fathers). The second one is that the education of genetically unrelated mothers does matter; that is, the positive nurturing effect of maternal education (we estimate on a sample of 157 egg donor children) is large enough to be statistically significant but not precise enough to statistically rule out effect sizes smaller than 0.025 (which is about one third of the overall intergenerational

 $^{^{20}}$ In our intergenerational mobility sample of rearing mothers of egg donor children, we are able to identify the likely donor for 62 percent of all donor children. We have replaced missing education with the average education of the donors treated in the same year and clinic, and added a missing dummy for those donor children without donor information to the intergenerational mobility model. Excluding the missing observations does not affect our results. When we estimate the intergenerational model in column 6 using the subsample deleting the missing observations, the coefficients on the mother's education, father's education, and the matched donor's education are 0.059 (with a standard error of 0.041), 0.075 (with a standard error of 0.038), and 0.005 (with a standard error of 0.041), respectively. While less precise, these estimates do not differ in any material way from those reported for the full sample.

association for nondonor mothers). The third one is that the education of genetically unrelated mothers matters more than that of genetically unrelated fathers. When we test for differential impacts of the unrelated fathers and mothers (by pooling the donor samples and estimate columns 4 and 5 with a fully interacted regression model), we find that the nurture impact of mothers is significantly larger than that of fathers (the estimated difference is equal to 0.080 and comes with a standard error of 0.026).

Specification Issues

We consider several specification issues that are common to the analysis of intergenerational skill transmission. We focus here on those regression models that take account of first treatment date controls and clinic fixed effects (as in columns 2, 4 and 5 of table 3) and examine how assortative mating, nonlinearities in intergenerational skill transfers, and the use of alternative parental education and child achievement measures affect the nurture effect estimates.

First, we estimate the nurture effects in intergenerational transmission models that control for the educational attainment, birth year, and citizenship of both parents. Our motivation is that such a specification takes account of the nurture and nature effects of the genetically related parent that are due to assortative mating. One concern is that we may inadequately control for assortative mating spillovers and that the nurture effect is biased capturing not only the nurture effect from the genetically unrelated parent but also part of the nurture and nature effects from the other genetically related parent. To check whether assortative mating spillovers are empirically relevant, we estimate nurture effects with the largest assortative mating spillover bias possible by excluding all the controls of the genetically related parent. Large nurture effects would then indicate that the corresponding nurture effects are sensitive to assortative mating spillovers. Table 4 (panel A) reports the corresponding intergenerational transmission estimates taken from regression models that control for the education, birth year, and citizenship of only one parent. We note here that the reported estimates attached to either the mother's and father's education come from separate regressions. As expected, we find that the intergenerational associations are (almost all) larger than the estimates in table 3, in particular, the nurture effect of the father's education on the child's test scores is now large enough to be statistically significant. While assortative mating spillovers generally produce upward biased nurture effect estimates, our findings mirror those

Table 4			
Alternative regressions	of child test	scores on their	parents' education

	non donor children	sperm donor children	egg donor children	
	(1)	(2)	(3)	
A. Alternative specification:	excluding spo	ousal characteris	tics	
years of education mother	$0.105 \\ 0.004^{***}$	$0.062 \\ 0.014^{***}$	$0.085 \\ 0.028^{***}$	
years of education father	0.096 0.002^{***}	0.023 0.011^{***}	0.088 0.026^{***}	
B. Alternative specification:	including inte	eractions		
years of education mother	0.081 0.013***	$0.035 \\ 0.066$	0.073 <i>0.175</i>	
years of education father	$0.070 \\ 0.013^{***}$	$\substack{-0.042\\0.080}$	0.065 <i>0.155</i>	
years of education interaction	$\substack{-0.0004\\0.001}$	$0.002 \\ 0.005$	$^{-0.0001}_{0.012}$	
C. Alternative independent	variable: colle	ge education		
college education mother	0.295 0.017***	0.238 0.051***	$0.165 \\ 0.137$	
college education father	$\begin{array}{c} 0.311 \\ 0.018^{***} \end{array}$	$-0.012 \\ 0.050$	0.444 0.133^{***}	
D. Alternative dependent va	riable: reading	g test scores		
years of education mother	0.075 0.006***	0.062 0.015^{***}	0.097 0.036***	
years of education father	0.054 0.007***	$\begin{array}{c}-0.001\\0.014\end{array}$	$0.042 \\ 0.030$	
E. Alternative dependent va	riable: math t	est scores		
years of education mother	$0.065 \\ 0.016^{***}$	0.072 0.027^{***}	$0.009 \\ 0.026$	
years of education father	$0.064 \\ 0.017^{***}$	$\substack{-0.028\\0.022}$	0.108 0.045***	

Note-The table contains estimates from specifications that deviate from the baseline regression models (reported in columns 2, 4 and 5 of table 3). Panel A: the main independent variable is either the mother's or the father's educational attainment measured in the nominal years spent in school. Panel B: the main independent variables are the educational attainment (measured in the nominal years spent in school) of the mother, father, as well as the interaction between the two. Panel C: the main independent variables are the parents' college education indicators measuring whether parents completed 15 or more years of education. Panel D: the dependent variable is the averaged standardized child test score in math. Specifications in other panels control for the gender and birth year of children, and Danish citizenship and birth year of both parents. All specifications include first IVF treatment date controls (measured in calender months) and a full set of clinic indicators. Standard errors are clustered by mothers and shown in italics; * indicates significance at 10 percent level, ** indicates significance at 5 percent level, and *** at 1 percent level.

obtained in earlier regressions. For genetically unrelated mothers, we find that the strong intergenerational associations remain essentially unchanged. For genetically unrelated fathers, we find that intergenerational associations remain quite small and significantly smaller than those obtained for genetically related fathers. There is thus little evidence that our nurture effects estimates pick up some of the nature effects of the genetically related partner.²¹

Second, we estimate the nurture effects in a simple intergenerational transmission model where child skills depends linearly (and additively) on the educational attainment of both parents. The concern is that such a model may miss important nonlinearities in the intergenerational transmission of human capital skills that are induced by, for example, child-rearing complementarities between mothers and fathers or nature-nurture interactions. We test for nonlinearities in intergenerational skill transmission by adding interactions between the educational attainment measures of mothers and fathers. Table 4 (panel B) contains these intergenerational transmission estimates (based on intergenerational models 5 and 6). We find that all the interaction estimates are small and statistically insignificant. Given that we get precisely estimated zeros in the larger samples of nondonor children, we consider nonlinear transmission channels unlikely.

Third, we measure parental education by the number of years parents spent in school and estimate how child test scores respond to a one year increase in parental education (net of genetic skill transfers). The concern here, which is related to the nonlinearity concern mentioned above, is that the test score gains of children may depend on whether parents spend one more year in elementary school, high school, or college. If, for instance, child test scores respond stronger to parental school years spent in lower secondary education than to parental school years spend in university, we may find weaker nurture effects because we estimate our intergenerational mobility regression models on samples where particularly better educated fathers are overrepresented. This would be a valid concern had we observed positive nurture effects in models with parental college indicators as the main independent variables. This is not the case. Table 4 (panel C) reports additional intergenerational transmission estimates for parental college education. College education is a dummy variable and indicates whether the parent has spent at least two years in college or more. With college

 $^{^{21}}$ Collado, Ortuño-Ortìn and Stuhler (2019) make a similar point when they model and quantify intergenerational transmission of education using extended family members (defined by common great grandparents) and their spouses: that is, partners hardly sort on genetic factors that drive their educational attainment.

education, our findings are consistent to those previously reported. The intergenerational associations between the genetically unrelated father and child, which we interpret as nurture effects, are statistically insignificant and close to zero. The intergenerational associations between the genetically unrelated mother and children are less precisely estimated but continue to be large.

And lastly, we measure the educational achievement of children by overall test score performance averaging the test score performance on nationwide reading and math tests. One concern is that the test score gains may depend on test subject. Table 4 (panels D and E) reports intergenerational mobility estimates for standardized reading and math test scores separately. We find that all the intergenerational associations are less precisely estimated because we work with smaller samples.²² We also find that the intergenerational associations (including those we interpret as the nurturing impacts of parental education) for reading and math test scores are very similar to the associations we estimate for the combined reading-math test scores. One exception is the relationship between the education of genetically unrelated mothers and child test scores in math, which is noticeably lower. But since the latter estimate for math is not precise enough to rule out an effect size as large as 0.060, it is difficult to draw firm conclusions about a nurturing effect of maternal education for child test scores in math, but not in reading.

7 Generalizability of Results

An important question is whether the nurture effect estimates taken from parents of donor children are (informative about and) generalizable to other parents of nondonor children. In an attempt to answer this external validity question, we compare intergenerational mobility patterns across different samples of parents and children.

IVF Families

We first compare IVF treated families with and without donor children. As we reported earlier, there are two external validity risks. The first one is that

 $^{^{22}}$ We lose precision for two reasons. First, the test score performance is based on fewer tests. Second, the samples for children with available math test scores are considerably smaller, consisting of 11,532 nondonor children, 560 sperm donor children, and 123 egg donor children. There are fewer children with math test scores because children do fewer math tests and take their first math test at a later age.

the nurture effects are heterogeneous: that is, nurture effects are different for different families. The second risk is one of treatment differentials: that is, parents treat donor children differently than nondonor (but otherwise similar) children.

To test for heterogeneous nurture effects, we focus on the rearing parents in donor and nondonor families and compare how the educational attainment of parents relates to the test scores of their genetically related children. For mothers of sperm donor and nondonor children, we find that increasing maternal education by a year is associated with standard deviation gains of 0.067 and 0.076 for child test scores (see table 3, columns 2 and 4). For fathers of egg donor and nondonor children, we find that one more year of education leads to comparable standard deviation gains of 0.063 and 0.064 for child test scores (see table 3, columns 2 and 5). With intergenerational associations being nearly identical, together with interaction associations being close to zero (see table 4, panel B), we find no clear evidence of heterogeneous nurture effects and conclude that families with donor children are comparable to families with nondonor children in terms of intergenerational skill transfers.

To test for treatment differentials, we focus on genetically related and unrelated parents in IVF families and compare their parental leave take up, labor supply, and divorce risk during the child's preschool years. Given that our nurture effect estimates are so different for fathers and mothers, we distill two possible external validity concerns. The concern with fathers of sperm donor children is that the zero nurture effects may arise because fathers feel less attached to their genetically unrelated children and take up less parental leave, work longer hours, and face higher divorce risks. And reversely, the concern with mothers of egg donor children is that the positive nurture effects may arise because they feel more attached to their children and take up more parental leave, work fewer hours, and face lower divorce risks.

Table 5 reports estimates from least-squares regressions on the sample of all IVF-treated families, with four different dependent variables: parental leave take up measured as the number of registered parental leave days taken by the parents during the first two and five years following child birth, labor supply response of parents measured as the percentage change between the average labor earnings in the four years before child birth and the average labor earnings in the first five years after child birth, and divorce measured as an indicator for whether married/cohabitating parents divorce/break up during the first five years following child birth. The independent variables of interest are indicators

	days _I leave ^a	oarental (0-2 yrs)	days _] leave ^a	parental (0-5 yrs)	labor change	supply (0-5 yrs)	divorce (0-5 yrs)
	mother (1)	father (2)	mother (3)	father (4)	mother (5)	father (6)	couple (7)
<i>vondonor families (ref.)</i> sperm donor families	0.506	1.240	-5.953	2.336	-0.043	-0.001	0.034
à	6.216	1.275	11.530	1.518	0.050	0.030	0.022
egg donor families	8.235	-4.598	-11.560	-4.559	0.035	0.055	-0.056
1	10.242	1.160^{***}	14.087	1.286^{***}	0.059	0.067	0.022^{***}
R-squared	0.173	0.031	0.145	0.031	0.026	0.028	0.012
number of observations	15,396	15,396	11,566	11,566	14,541	14,239	15,396
mean (nondonor families)	285.203	15.937	346.546	18.554	-0.047	0.138	0.165

Table 5 Relationship between parents' time spent with children and donor type in IVF treated families

days during the first two and five years following child birth (columns 1 to 4), labor supply response measured as the percentage change between the average labor earnings in the four years before child birth and the average labor earnings in the first five years after child birth (columns 5 and 6), and divorce measured as indicator for whether married/cohabitating parents divorce/break up during the first five years following child birth (column 7). The independent variables of interest are indicators for egg and sperm donor families. The nondonor families serve as the reference group. The samples treatt each donor family as separate observation. All specifications control for the child's gender and birth year, the parents' critizenship and birth year, the first treatment date (in calendar months) and a full set of clinic indicators. Robust standard errors are shown in italics, "indicates significance at 10 percent level, ** indicates for whether nevel, and *** at 1 percent level. (0 Before 2002, the paid parental leave policies (in our study period) allowed parents to take up 70 days maternity leave, 10 days paternity leave and 50 days shared leave. After 2002, these paid parentity leave, 10 days maternity leave, 10 days paternity leave and 50 days shared leave. Note-The dependent variables are the parents' parental leave take up measured as the number of registered parental leave

for egg and sperm donor families. As before, we include (but not report on) pre-determined controls for the child's gender and birth year, and the parents' citizenship and birth year in our regressions. Table 5 also reports summary statistics of parental leave take up, change in labor supply, and divorce rates. The sample sizes vary because of missing earnings averages and because of unavailable 5-year parental leave measures for the youngest cohorts of children.

Two things become clear from this table. First, the parental leave statistics clearly show that all fathers spend much less time with their children than mothers. Second, there is little evidence of the treatment differential concerns we expressed above. The estimated coefficients on sperm donor families are all statistically insignificant and mostly small, suggesting that fathers of sperm donor children spend as little time with their children as fathers of nondonor children. Also the estimated coefficients on egg donor families indicate that mothers of egg donor children spend not more but as much time with their children as mothers of nondonor children do. Only the divorce coefficients are sizable and have signs in the hypothesized directions; that is, compared to nondonor couples, we find that egg donor couples are 6 percentage points less likely to divorce, whereas sperm donor couples are 3 percentage points more likely to divorce. It is unlikely, though, that such divorce patterns alone can be held responsible for the nurturing impacts observed in this study. For mothers, the corresponding divorce estimate is large but not large enough to mask substantial nurture effects. And for fathers, the corresponding divorce estimate is not statistically significant and too small to explain away the zero nurture effects. As an additional check, we have run our intergenerational mobility regressions on a sample of arguably more attached fathers, deleting all the divorced couples from the sample. With a point estimate of -0.011 (0.016) with standard error in parentheses, we continue to find a zero nurture effect for nondivorced rearing fathers of sperm donor children.

We view these test results as supportive evidence of the greater external validity of the nurture effect estimates within our population of successfully treated IVF families.

Representative Families

We next compare IVF families and representative families. In particular, we compare intergenerational mobility patterns between IVF families with genetically related children and families drawn from the full population of families with children born in the same research window. In both types of families, the intergenerational transmission estimates represent an overall blend of nature and nurture influences. If we would get the same intergenerational transmission estimates, we conjecture that the process of skill transmission is comparable in the two types of families, and that the intergenerational transmission estimates obtained with IVF families have a wider generalizability.

Table 6 presents the intergenerational transmission estimates for education for the different samples. In column 1 we report estimates for the sample of representative families. In column 2 we reproduce our baseline estimates for the sample of nondonor IVF families (as reported in table 3) for ease of comparison. We find that the estimated associations between parental schooling and child test scores are all large and positive. Although they are not identical, the intergenerational associations differ only a little across the two types of families.

Adoptive Families

And lastly, we compare intergenerational mobility patterns in families with donor children and families with adopted children. In both types of families, we can estimate nurture effects, that is, if children are genetically unrelated to their rearing parents. The adoptees are suitable for identifying (and comparing) nurture effects: adoptees are adopted at infancy (the vast majority of adoptees is adopted below the age of 2) to ensure that the nurture effect estimates capture most of the early childhood influences; and adoptees are foreign born to ensure that the assignment process is fairly random.²³

In columns 3 and 4 we report the intergenerational transmission estimates for foreign-born adoptees as well as for Korean-Danish adoptees. In columns 5 and 6 we reproduce our baseline estimates for the sample of donor children (as reported in table 3) for ease of comparison. For the sample of foreign-born adoptees, we find small, positive, and statistically significant nurture effects indicating that higher educated parents provide a better nurturing environment for adopted children to perform well in school. The associations are similar for mothers and fathers and imply that four more years of parental education of either parent are associated with children having about 0.10 standard deviation

 $^{^{23}}$ As discussed in Holmlund, Lindahl, and Plug (2011), most adoptive parents know little, if anything, about the biological background of foreign-born adoptees. They know, like we do, the adoptees' gender, age, and country of origin. We therefore run our intergenerational mobility regressions with additional controls for the adoptees' country of birth (measured by indicator variables for the 20 most popular countries of birth) and assume that assignment of adoptees to families is conditionally random.

	all other children	IVF nondonor children	all adopted children	Korean adopted children	sperm donor children	egg donor children
	(1)	(2)	(3)	(4)	(5)	(9)
years of education mother	0.085 0.001***	0.076 0.078***	0.019 0.019	0.023 0.023	0.067 0.018***	0.072 0.001***
years of education father	$0.072 \\ 0.001^{***}$	0.003^{***}	0.026 0.008^{***}	0.020 0.022	$-0.008 \\ 0.015$	0.022^{***}
R-squared	0.162	0.132	0.127	0.054	0.085	0.258
number of children	650,930	19,509	2,674	232	820	157
country-of-origin FE clinic FE, calendar month		>	>		>	>

	education
	parents'
	their
	uo
	scores
	\mathbf{test}
	child
	\mathbf{of}
	regressions
	validity
Table 6	External

Note-The dependent variables are averaged standardized test scores. The independent variables are the parents' educational attainment measured in the nominal years spent in school. All specifications control for the gender and birth year of children, and the birth year and Danish citizenship of mothers and fathers. The specification in column 3 further controls for country-of-origin fixed effect (for the 20 most popular countries of origin). The specifications in column 2, 5 and 6 further controls for clinic fixed effects and calendar month of first donor treatment. Standard errors are clustered by mothers and shown in trailis, * indicates significance at 10 percent level, ** indicates significance at 5 percent level, and *** at 1 percent level.

higher test scores. For the sample of Korean-Danish adoptees, the estimates are also positive and small, but too imprecise to make informative comparisons.

How are we to interpret the difference between the nurture effect estimates taken from adopted children to those taken from donor children? If we think of the nurture effect estimates obtained with adoptive parents and foreign-born adoptees as representative for all other parents and children (including IVF parents and donor children), the difference should capture the nurturing influence of parental education on the prenatal and very early childhood environment. We find the largest difference for rearing mothers, that is, the intergenerational transmission estimate for mothers of egg donor children is three times as high as the estimate for mothers of adopted children. When we test for differential impacts (by pooling the egg donor and adoption samples and estimate columns 3 and 6 with a fully interacted regression model), we find that the nurture impact of egg donor mothers is significantly larger than that of adoptive mothers (the estimated difference is equal to 0.052 and comes with a standard error of (0.023). These nurture effect estimates provide some suggestive evidence that better educated mothers are better able in creating a prenatal and early childhood environment that improves child test scores.

We find the opposite, but smaller, difference for rearing fathers, that is, the intergenerational transmission estimate appears somewhat larger for fathers of adopted children than for fathers of sperm donor children. An interpretation is difficult because the estimated difference is not precise enough to reject that the two nurture effect estimates are different. Perhaps there is no difference. Or perhaps there is a marginally larger nurture effect for fathers of foreign-born adoptees, which may then be attributed to the selective screening of prospective adoptive parents. Over the period we consider, prospective parents were assessed on the basis of formal criteria (regarding background, age, marital stability, income et cetera) and in-depth interviews (regarding personal history and family relationships) before they were qualified to adopt. If selected prospective adoptive parents have somewhat stronger parenting skills than the average IVF parents in our sample, selective screening may explain the marginally stronger impact for adoptive fathers but at the same time may compromise (and underestimate) the prenatal and early childhood environment interpretation of the differential impact of donor and adoptive mothers.

Interpreting the Results

Our findings clearly suggest that, once we take account of genes, better educated fathers are of little help to their children's test scores, while better educated mothers raise children with better test scores. In particular, when we take our estimates at face value, it seems that fathers impact children's test scores only through genes and mothers impact children's test scores only through environment. What is the explanation?

On the one hand, we are surprised to find that the genes of fathers are so much more important than those of mothers. If genes are passed on to children from fathers and mothers alike, we would expect that mothers impact their children's test scores also through genes, and that their environmental influences would augment their genetic contribution, not replace it. We have explored several possible explanations, including donor mothers spending more time with their donor children (they don't, see table 5), donor children having a genetic advantage over nondonor children (they haven't, see table 1), and the presence of nature-nurture interactions (there aren't, see table 4).²⁴ While we can rule out a few explanations, some remain. Perhaps better educated donor mothers have better child-rearing skills, which mask the mothers' genetic impact on child test scores. Or perhaps the donor samples are just too small to uncover that mothers impact child test scores through genes (and fathers through environment).

On the other hand, we are not surprised to find that mothers contribute more to the child's environment than fathers do. A large part of the explanation must be this: prenatal (and very early childhood) conditions are crucial for the development of child skills. It explains why the nurture effects in donor families are much stronger for mothers than for fathers: unlike fathers, mothers are pregnant, carry children, give birth, and spend observably more time taking

²⁴To motivate these explanations, we focus on mothers and decompose their intergenerational coefficients β^m and α^m into a nature component (h), nurture component (e), and the interaction between the two $(h \times e)$: $\beta^m = h^m + e^m + h^m \times e^m$, and $\alpha^m = e^{dm} + h^{ed} \times e^{dm}$, where superscripts m, dm, and ed stand for nondonor mothers, donor mothers (donor recipients), and egg donors (donor providers), respectively. Taking the estimated coefficients at face value, we set β^m and α^m equal to each other to arrive at the following nature expression $h^m = [e^{dm} - e^m] \times [1 + h^{ed}] + [h^{ed} - h^m] \times e^m$. It is easy to see that we underestimate the influence of genes in case (i) donor mothers have better child-rearing skills and/or provide a more advantageous environment for their children $(e^{dm} > e^m)$; and (ii) donor mothers raise children with certain genetic advantages $(h^{ed} > h^m)$ and exhibit nature-nurture interactions. We consider the latter channel unlikely, given that egg donors have no educational advantage over other mothers (see table 1) and interaction proxies are small and statistically insignificant (see table 4).

rtegressions of prenatal shloking	, on equeation	using IVI treat	ea women
	non donor treatment	sperm donor treatment	egg donor treatment
	(1)	(2)	(3)
years of education treated woman	$-0.019 \\ 0.002^{***}$	$^{-0.025}_{0.004^{stst}}$	$-0.014 \\ 0.007^{*}$
R-squared	0.030	0.047	0.031
number of treated women	74,904	6,443	589
mean prenatal smoking	0.121	0.117	0.093

 Table 7

 Regressions of prenatal smoking on education using IVF treated women

Note-The dependent variable is prenatal smoking (0/1). The main independent variable is the (prospective) mother's educational attainment measured in the nominal years spent in school. The samples treat each IVF attempt as separate observation. All specifications further control for the partner's educational attainment, Danish citizenship and birth year of both partners, first IVF treatment date controls (measured in calender months) and a full set of clinic indicators. Standard errors are clustered by prospective mothers and shown in italics; * indicates significance at 10 percent level, ** indicates significance at 5 percent level, and *** at 1 percent level.

care of children in early childhood. And it explains why the nurture effects for rearing mothers are much stronger for donor children than for adopted children: unlike adopted children, donor children benefit from their mother's exposure in pregnancy and very early childhood.

We next take a closer look at the relationship between prenatal environment and the education of (prospective) mothers. In particular, we have prenatal information on whether IVF treated women smoke during their pregnancy, recorded in the IVF register for the period 2006 to 2011, which enables us to regress a prenatal smoking indicator on the years of education of both partners (in the treated couple) and the standard controls including the partners' citizenship and birth year, the calender month of treatment and clinic fixed effects. We consider all treated women including unsuccessfully treated women to account for possible prenatal smoking influences on the success of treatment (having a livebirth). If better educated women are better in avoiding prenatal risks, we should find that better educated women are less likely to smoke during pregnancy.

Table 7 reports the least-squares regression results using the sample of all IVF women who have ever been treated between the years 2006-2011. We find that the estimates attached to the treated woman's years of schooling are

all statistically significant, negative and sizable, suggesting indeed there is less prenatal smoking among all better educated pregnant women, including those who carry genetically unrelated babies.

8 Discussion and Conclusion

In this paper we investigate the intergenerational persistence in human capital skills, net of genetic skill transfers, using a novel strategy based on Danish children conceived through sperm and egg donations in IVF treatments. By considering donor children, we can eliminate the genetic connection between children and one of their parents, that is, children from sperm donors are genetically related to their mother but not their father, and children from egg donors are genetically related to their father but not their mother. We measure intergenerational persistence in human capital skills by estimating the relationship between parental education and child test performance on nationwide tests taken in primary and secondary education using samples of genetically related and genetically unrelated children.

We find, first, that there is a strong relationship between parental education and child test scores for donor-treated parents and their genetically related children. The intergenerational associations, which represent a blend of nurture and nature effects, are all large and positive, and similar to the ones we estimate using a conventional representative sample of parents and children. Second, we find an equally strong human capital relationship for donor-treated mothers and their genetically unrelated children, suggesting that most of the human capital skill transmission between mothers and children is nurture driven. The intergenerational associations are statistically significant and large, as large as the ones we find for mothers of genetically related children, and significantly larger than the ones we find for mothers of adopted children. And third, we find that there is no such relationship for donor-treated fathers and their genetically unrelated children, suggesting that the strong associations between fathers and their genetically related children are mostly nature driven. The intergenerational associations are practically zero and significantly smaller than the ones we find for fathers of genetically related children as well as the ones we find for donor-treated mothers of genetically unrelated children.

Of course, our empirical strategy only works if donor assignment is either random or related to variables that we observe and control for. In the Danish context, we know exactly how donors are assigned to prospective parents. For rearing mothers of egg donor children, the assignment of eggs is based on the position mothers take on the clinic's waiting list. With regression models that account for the clinic queue order, we show that donor assignment is as good as random. For rearing fathers of sperm donor children, the assignment of donor sperm is based on unobserved donor preferences. If the assignment of sperm donors is selective, our primary concern would be that part of what we interpret as the nurture effect may in fact be genetic. With nurture estimates for fathers close to zero, however, we consider such a selection bias unlikely. Because the assignment of donors to parents is not selective, we can interpret the nurture effect estimates (as presented for mothers and fathers) in a causal way and conclude that it is the nurturing impact of the education of mothers that matters most for the test scores of children.

What could be the mechanism behind these findings? While keeping in mind the standard caveats about interpreting reduced-form findings, our nurture effect estimates are consistent with the notion that prenatal and early childhood conditions are essential for the development of child skills. It explains why mothers matter more than fathers: unlike fathers, mothers are the ones who carry children, give birth, and observably spend much more time taking care of children during early childhood. It explains why donor mothers matter more than adoptive mothers: unlike mothers of adopted children, mothers of egg donor children are the ones who provide their children's prenatal and very early childhood environment. And it explains why high educated mothers matter more than low educated mothers: with more human capital skills, mothers are better in avoiding prenatal risks (in terms of reduced smoking in pregnancy) and spend more quality time with their children during their very early childhood years (as documented in the time use studies of Guryan, Hurst, and Kearney (2008) and Bonke (2009)).

To the eye, our finding that education of mothers matters more than that of fathers, once we take account of genetic skill transfers, appears contrary to those of recent adoption studies (Plug 2004, Björklund, Lindahl, and Plug 2006, Sacerdote 2007, Fagereng, Mogstad, and Rønning 2018, Black et al. 2019). This is not so surprising, however, given that we use a different strategy and study different outcomes that are realized in a different country. Our study relies on donor children. The other studies rely on adoptees. With rearing mothers of adoptees, for example, it is not only possible but (as we have shown) highly probable to get weaker nurture effect estimates because the adoption strategy misses those beneficial influences of maternal education on the child's prenatal and very early environment that our strategy with rearing mothers of donor children captures. Also our study examines test score outcomes in primary and lower secondary education when children are young and education is mandatory. The adoption studies examine skill outcomes of children realized in (early) adulthood including overall educational attainment, earnings, and wealth. With different outcomes, it is possible to get larger nurture effects for fathers if better educated fathers matter more for later-life outcomes that are more financially intensive and less time intensive, such as attending university, labor market outcomes, or portfolio holdings. In addition to this, our study examines intergenerational persistency in human capital skills in Denmark where education is heavily subsidized, skill returns are low, and parental leave arrangements are generous. Some adoption studies look at intergenerational persistency in the US (Plug 2004, Sacerdote 2007). With children growing up in countries with more costly education and higher skill returns, it is possible to get larger nurture effects for fathers if better educated fathers (who would otherwise see little return to their child investments) devote more time and money on their children. And with children growing up in countries with less generous parental leave arrangements, it is also possible to get weaker nurture effects for mothers if better educated mothers (who would otherwise spend more time with their children) take up less maternity leave. Clearly, much more work needs to be done to explore these possibilities; this is a priority for our future research.

References

- Almond, Douglas, Janet Currie, and Valentina Duque. 2018. "Childhood Circumstances and Adult Outcomes: Act II," Journal of Economic Literature, 56:4 pp. 1360-1446.
- Amin, Vikesh, Petter Lundborg, and Dan-Olof Rooth. 2015. "The Intergenerational Transmission of Schooling: Are Mothers Really Less Important Than Fathers?" *Economics of Education Review*, 47, pp. 100-117.
- Ashenfelter, Orley, and Alan B. Krueger. 1994. "Estimates of the Returns to Schooling from a New Sample of Twins," *American Economic Review*, 84:5, pp. 1157-1173.
- Behrman, Jere R., and Mark R. Rosenzweig. 2002. "Does Increasing Women's

Schooling Raise the Schooling of the Next Generation?" American Economic Review, 92:1, pp. 323-334.

- Bingley, Paul, Kaare Christensen, and Vibeke Myrup Jensen. 2009. "Parental Schooling and Child Development: Learning from Twin Parents," Danish National Centre for Social Research Social Policy and Welfare Services Working Paper 07:2009, Copenhagen.
- Björklund, Anders, Markus Jäntti and Gary Solon. 2005. "Influences of Nature and Nurture on Earnings Variation: A Report on a Study of Various Sibling Types in Sweden," in Unequal Chances: Family Background and Economic Success, Samuel Bowles, Herbert Gintis and Melissa Osborne, eds. Princeton University Press.
- Björklund, Anders, Mikael Lindahl, and Erik Plug. 2006. "The Origins of Intergenerational Associations: Lessons from Swedish Adoption Data," *Quarterly Journal of Economics*, 121:3, pp. 999-1028.
- Björklund, Anders and Kjell Salvanes. 2011. "Education and Family Background: Mechanisms and Policies," in *Handbook of the Economics of Education.* Eric A. Hanushek, Stephen Machin, and Ludger Woessmann, eds. Vol. 3, pp. 201-247. Elsevier.
- Black, Sandra and Paul Devereux. 2011. "Recent Developments in Intergenerational Mobility," in *Handbook of Labor Economics*. Orley Ashenfelter and David Card, eds. Vol. 4, pp. 1487-1541. Elsevier.
- Black, Sandra, Paul Devereux and Kjell Salvanes. 2005. "Why the Apple Doesn't Fall Far: Understanding Intergenerational Transmission of Human Capital," *American Economic Review*, 95:1, pp. 437-449.
- Black, Sandra, Paul Devereux, Petter Lundborg, and Kaveh Majlesi. 2019. "Poor Little Rich Kids? The Role of Nature versus Nurture in Wealth and Other Economic Outcomes and Behaviors," *Review of Economic Studies*, forthcoming.
- Bonke, Jens. 2009. Forældres Brug af Tid og Penge på deres Børn. Parental Time and Money Spent with Children. The Rockwool Foundation Research Unit, Copenhagen, Denmark.

- Brandén, Gunnar, Mikael Lindahl, and Björn Öckert. 2017. "The Importance of Nature-Nurture Interactions in Skill Formation: Evidence from a Large Sample of Adoptees," University of Gothenburg, mimeo.
- Brodzinsky, David. 1987. "Adjustment to Adoption: A Psychosocial Perspective." Clinical Psychology Review, 7:1, pp. 25-47.
- Case, Anne, I-Fin Lin, and Sara McLanahan. 2000. "How Hungry is the Selfish Gene?" *Economic Journal*, 110:466, pp. 781-804.
- Case, Anne, I-Fin Lin, and Sara McLanahan. 2001. "Educational Attainment of Siblings in Stepfamilies" *Evolution and Human Behavior*, 22:4, pp. 269-289.
- Cesarini, David. 2010. Essays on Genetic Variation and Economic Behavior. PhD Thesis. Massachusetts Institute of Technology.
- Cesarini, David and Peter Visscher. 2017. "Genetics and Educational Attainment," *npj Science of Learning*, 2.1: 4.
- Ciconia (2015). Patient information (Egg donation). Aarhus Fertilitetsklinik.
- Collado, Dolores, Ignatio Ortuño-Ortin, and Jan Stuhler. 2019. Estimating Intergenerational and Assortative Processes in Extended Family Data. Working Paper. Universidad Carlos III de Madrid.
- Cunha, Flavio, and James Heckman. 2007. "The Technology of Skill Formation," American Economic Review, 97:2, pp. 31-47.
- Dawkins, Richard. 2006. The Selfish Gene: With a New Introduction by the Author. UK: Oxford University Press.(Originally published in 1976).
- Dearden, Lorraine S., Stephen Machin, and Howard Reed. 1997. "Intergenerational Mobility in Britain," *Economic Journal*, 110:440, pp. 47-64.
- Fagereng, Andreas, Magne Mogstad, and Marte Rønning. 2018. "Why Do Wealthy Parents Have Wealthy Children?" CESifo Working Paper, No. 6955, Center for Economic Studies and Ifo Institute (CESifo), Munich. Forthcoming Journal of Political Economy.
- Guryan, Jonathan, Erik Hurst, and Melissa Kearney. 2008. "Parental Education and Parental Time with Children," *Journal of Economic Perspectives*, 22:3, pp. 23-46.

- Griliches, Zvi. 1979. "Sibling Models and Data in Economics: Beginnings of a Survey," *Journal of Political Economy*, 87:5, pp. 37-64.
- Holmlund, Helena, Mikael Lindahl and Erik Plug. 2011. "The Causal Effect of Parents' Schooling on Children's Schooling: A Comparison of Estimation Methods," *Journal of Economic Literature*, 49:3, pp. 615-51.
- Hægeland, Torbjørn, Lars Kirkeboen, Oddbjørn Raaum and Kjell Salvanes. 2010. Why Children of College Graduates Outperform their Schoolmates: A Study of Cousins and Adoptees. IZA Discussion Paper No. 5369, Bonn, Germany.
- Jencks, Cristopher, and Marsha Brown. 1977. "Genes and Social Stratification," in *Kinometrics: The Determinants of Economic Success within* and between Families. Paul Taubman, eds. Amsterdam: North-Holland Publishing Company, pp. 169-231.
- Isacsson, Gunnar. 1999. "Estimates of the Return to Schooling in Sweden from a Large Sample of Twins," *Labour Economics*, 6:4, pp. 471-89
- Larsen, Elisabeth Clare, Karsten Petersen, Anders Nyboe Andersen and Søren Ziebe. 2009. "Ægdonation i Danmark (Egg Donation in Denmark)," Ugeskrift Læger 171:39, pp. 2822-2825.
- Lundborg, Petter, Erik Plug and Astrid Würtz Rasmussen. 2017. "Can Women have Children and a Career? IV Evidence from IVF Treatments," *American Economic Review*, 107:6, pp. 1611-1637.
- Lichtenstein, Paul, Nancy L. Pedersen, and Gerald E. McClearn. 1992. "The Origins of Individual Differences in Occupational Status and Educational Level: A Study of Twins Reared Apart and Together," Acta Sociologica, 35:1, pp. 13-31.
- Lykken, David T., Thomas J. Bouchard, Matt McGue, and Auke Tellegen. 1990. "The Minnesota Twin Family Registry: Some Initial Findings," Acta Geneticae Medicae et Gemellologiae: Twin Research, 39:1, pp. 35-70.
- Miller, Paul, Charles Mulvey, and Nick Martin. 1995. "What Do Twins Studies Reveal about the Economic Returns to Education? A Comparison of Australian and US Findings," *American Economic Review*, 85:3, pp. 586-599.

- Beuchert, Louise V., Anne B. Nandrup. 2018. "The Danish National Tests at a Glance," *Nationaloekonomisk Tidsskrift*, 1, pp. 1-37.
- Nielsen, Anders Faurskov, Bjørn Pedersen, and Jørgen Glenn Lauritsen. 1995. "Psychosocial Aspects of Donor Insemination: Attitudes and Opinions of Danish and Swedish Donor Insemination Patients to Psychosocial Information Being Supplied to Offspring and Relatives," Acta Obstetricia et Gynecologica Scandinavica, 74:1, pp. 45-50.
- Plomin, Robert John C. DeFries, and Gerald E. McClearn. 1990. Behavioral Genetics: A Primer. 3d ed. New York: W.H. Freeman, 1990.
- Plomin, Robert, John DeFries, and David Fulker. 1988. Nature and Nurture During Infancy and Early Childhood. Cambridge, UK: Cambridge University Press.
- Plug, Erik, and Wim Vijverberg. 2003. "Schooling, Family Background, and Adoption: Is it Nature or is it Nurture?" Journal of political Economy, 111:3, pp. 611-641.
- Plug, Erik. 2004. "Estimating the Effect of Mother's Schooling on Children's Schooling Using a Sample of Adoptees," *American Economic Review*, 94:1, pp. 358-68.
- Pronzato, Chiara. 2012. "An Examination of Paternal and Maternal Intergenerational Transmission of Schooling," *Journal of Population Economics*, 25:2, pp. 591-608.
- Rice, Frances, Gordon T. Harold, Jacky Boivin, Dale F. Hay, Marianne van den Bree, Anita Thapar, and Gene E. Robinson. 2009. "Disentangling Prenatal and Inherited Influences in Humans with an Experimental Design." *Proceedings of the National Academy of Sciences*, 106:7, pp. 2464-2467.
- Rutter, Michael. 2006. Genes and Behavior: Nature-Nurture Interplay Explained. Oxford: Blackwell Publishing.
- Sacerdote, Bruce. 2000. "The Nature and Nurture of Economic Outcomes," National Bureau of Economic Research, working paper 7949.
- Sacerdote, Bruce. 2002. "The Nature and Nurture of Economic Outcomes," American Economic Review (Papers and Proceedings), 92:2, pp. 344-48.

- Sacerdote, Bruce. 2007. "How Large Are the Effects from Changes in Family Environment? A Study of Korean American Adoptees," *Quarterly Journal* of Economics, 122:1, pp. 119-157.
- Sacerdote, Bruce. 2011. "Nature and Nurture Effects on Children's Outcomes: What Have We Learned from Studies of Twins and Adoptees?" in *Handbook of Social Economics*. Jess Benhabib, Alberto Bisin, and Matthew Jackson, eds. Vol. 1, pp. 1-30.
- Scarr, Sandra and Richard A. Weinberg. 1978. "The Influence of Family Background on Intellectual Attainment," *American Sociological Review*, 43:5, pp. 674-92.
- Scarr, Sandra and Richard A. Weinberg. 1994. "Educational and Occupational Achievements of Brothers and Sisters in Adoptive and Biologically Related Families," *Behavior Genetics*, 24:4, pp. 301-325.
- Sundhedsstyrelsen. 2012. Vejledning om Autoriserede Sundhedspersoners og Vævscentres Virksomhed og Forpligtelser i Forbindelse med Kunstig Befrugtning. (Guide for Authorized Health Professionals at Artificial Fertilization Related Working Environments) København: Sundhedsstyrelsen.
- Taubman, Paul. 1976. "The Determinants of Earnings: Genetics, Family, and Other Environments: A Study of White Male Twins," *American Economic Review*, 66:5, pp. 858-870.
- Teasdale, T. W., and David R. Owen. 1984. "Heredity and Familial Environment in Intelligence and Educational Level — A Sibling Study," *Nature*, 309:5969, pp. 620.
- Westergaard, Hanne Brix, Anne Mette Tranberg, Karin Erb, and Anders Nyboe Andersen. 2000. "Danish National IVF Registry 1994 and 1995. Treatment, Pregnancy Outcome and Complications During Pregnancy. Acta Obstetricia et Gynecologica Scandinavica, 79:5, pp. 384-389.
- Woessmann, Ludger. 2018. "Central Exit Exams Improve Student Outcomes," IZA World of Labor, 419.

Appendix Table A	
National test subjects by grade in primary and lowe	r secondary education

grade:	1	2	3	4	5	6	7	8	9
Danish, reading math English geography physics/chemistry biology		\checkmark	V	√		√ √	\checkmark	\checkmark	

_

Appendix Table B Variable descriptions

standardized test scoreTest score from Danish National Tests. The score is standardized to mean 0 and standard deviation 1 across each cohort. The math tests cover numbers and algebra, geometry, and applied mathe- matics. The reading tests cover language comprehension, decod- ing, and reading comprehension. Child gender is 1 for male, 0 for female. Child gender is 1 for male, 0 for female. (hild birth year years of schooling father college education motherwears of schooling father college education motherYear of birth of the child. Years of schooling (highest completed) of the mother. Years of schooling (highest completed) of the father. College education is 1 if mother has more than 14 years of com- pleted education, and 0 otherwise.college education fatherCollege education is 1 if father has more than 14 years of com- pleted education, and 0 otherwise.birth year mother birth year fatherYear of birth of the mother. Year of birth of the father.Danish citizenship mother change in annual earnings mother parental leave motherMother is of Danish origin, i.e. at least one of his parents is of Danish origin.parental leave mother parental leave fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave father parental leave fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave father parental leave fatherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years o	Variable	Description
maleChild gender is 1 for male, 0 for female.child birth yearYear of birth of the child.years of schooling motherYears of schooling (highest completed) of the mother.years of schooling fatherYears of schooling (highest completed) of the father.college education motherCollege education is 1 if mother has more than 14 years of completed education, and 0 otherwise.college education fatherCollege education is 1 if father has more than 14 years of completed education, and 0 otherwise.birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's life, and 0 otherwise.	standardized test score	Test score from Danish National Tests. The score is standardized to mean 0 and standard deviation 1 across each cohort. The math tests cover numbers and algebra, geometry, and applied mathe- matics. The reading tests cover language comprehension, decod- ing, and reading comprehension
IndeeControl generationchild birth yearYear of birth of the child.years of schooling motherYears of schooling (highest completed) of the mother.years of schooling fatherYears of schooling (highest completed) of the father.college education motherCollege education is 1 if mother has more than 14 years of completed education, and 0 otherwise.college education fatherCollege education, and 0 otherwise.college education fatherYear of birth of the mother.birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	male	Child gender is 1 for male 0 for female
control of the of other of the or bits of other of the or bits of other of the or bits of the o	child birth year	Vear of birth of the child
years of schooling father college education motherYears of schooling (highest completed) of the father. College education is 1 if mother has more than 14 years of com- pleted education, and 0 otherwise.college education fatherCollege education is 1 if father has more than 14 years of com- pleted education, and 0 otherwise.birth year motherYear of birth of the mother. Year of birth of the father.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	vears of schooling mother	Years of schooling (highest completed) of the mother.
college education motherCollege education is 1 if mother has more than 14 years of completed education, and 0 otherwise.college education fatherCollege education is 1 if father has more than 14 years of completed education, and 0 otherwise.birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave fatherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	years of schooling father	Years of schooling (highest completed) of the father.
College education fatherpleted education, and 0 otherwise.college education fatherCollege education is 1 if father has more than 14 years of completed education, and 0 otherwise.birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	college education mother	College education is 1 if mother has more than 14 years of com-
college education fatherCollege education is 1 if father has more than 14 years of completed education, and 0 otherwise.birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	0	pleted education, and 0 otherwise.
birth year motherpleted education, and 0 otherwise.birth year fatherYear of birth of the mother.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish citizenship fatherDanish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	college education father	College education is 1 if father has more than 14 years of com-
birth year motherYear of birth of the mother.birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish origin.Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	-	pleted education, and 0 otherwise.
birth year fatherYear of birth of the father.Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish citizenship fatherDanish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	birth year mother	Year of birth of the mother.
Danish citizenship motherMother is of Danish origin, i.e. at least one of her parents is of Danish citizenship fatherDanish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	birth year father	Year of birth of the father.
Danish citizenship fatherFather is of Danish origin, i.e. at least one of his parents is of Danish origin.change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	Danish citizenship mother	Mother is of Danish origin, i.e. at least one of her parents is of Danish origin.
change in annual earnings motherDifference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	Danish citizenship father	Father is of Danish origin, i.e. at least one of his parents is of Danish origin.
change in annual earnings fatherDifference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	change in annual earnings mother	Difference in mother log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).
parental leave motherTotal days of maternal leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.parental leave fatherTotal days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007.divorceDivorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	change in annual earnings father	Difference in father log average labor incomes 4 years before birth and 5 years after (missing income is counted as zero-income).
parental leave father Total days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007. divorce Divorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	parental leave mother	Total days of maternal leave two/five years after child's year of hirth (hirth year is year 1). Last year included is 2007
divorce Divorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.	parental leave father	Total days of parental leave two/five years after child's year of birth (birth year is year 1). Last year included is 2007
	divorce	Divorce is 1 if the couple divorced within the first five years of the child's life, and 0 otherwise.